INTEGRATE A HETEROGENEOUS SET OF DATA TOWARD DEVELOPING A FOREST HEALTH INDEX: A REVIEW

Rayan Binlajdam Cardiff University, UK BinlajdamR@cardiff.ac.uk Oktay Karakus Cardiff University, UK Karakus0@cardiff.ac.uk Omer Rana Cardiff University, UK RanaOF@cardiff.ac.uk

Pablo Orozco-terWengel Cardiff University, UK Orozco-terWengelPA@cardiff.ac.uk

Adisorn Lertsinsrubtavee Asian Institute of Technology, Thailand adisorn@ait.asia

Deepak Mishra University of New South Wales, Australia d.mishra@unsw.edu.au Benoit Goossens Cardiff University, UK GoossensBR@cardiff.ac.uk

Preechai Mekbungwan Asian Institute of Technology, Thailand preechaim@ait.asia

> **Charith Perera** Cardiff University, UK pererac@cardiff.ac.uk

February 5, 2024

ABSTRACT

Forest health monitoring has become a global issue after increased fire events, droughts, and tree diseases. Researchers developed techniques and approaches for determining a forest health index (FHI). FHI is a tool that uses one or more indicators to measure and assess the different aspects of a forest ecosystem, and it can vary depending on the goal of the assessment. Indicators in forest health refer to ecological, meteorological, and other indicators, such as biological ones. Ecological indicators include productivity, vitality, and biodiversity, while meteorological indicators include temperature, humidity, and precipitation. Different techniques and methodologies have been developed for measuring one or a group of these indicators. Remote sensing, field observation, and ground-based sensor approaches create a forest health monitoring system by monitoring indicators and generating indexes. Remote sensing, such as satellites, manned aircraft, and drones, is excellent for monitoring the forest's health status. However, relying on remote sensing alone to build a forest health monitoring system is still unreliable. This review aims to define forest health and the FHI while identifying various indicators and their relationship to creating an FHI. The paper describes several measurement approaches for developing a forest health monitoring system and defines several applications. The paper also examines the potential multi-data sources in developing a forest health monitoring system. The result suggests that integrating different techniques could enhance forest health monitoring.

Keywords Forest Health Index, Forest Health Monitoring, Remote Sensing, Ground-based Sensor, Field Observation, Multi-data Sources, UAV

1 INTRODUCTION

1.1 Motivation and Contributions

Forest health is a term used without a generally accepted definition. Indeed, forest health cannot be defined unambiguously as it reflects values from social, economic, and ecological aspects. Lausch et al. [1] covered several definitions of forest health. Trumbore et al. [2] quoted one of these definitions: a "mosaic of successional patches representing all stages of the natural range of disturbance and recovery." According to the United Nations (UN), "more than 80% of all land mammals, birds, insects, and plants inhabit forests. About one-third of humanity is estimated to depend on forests and their products directly."

The United States of America (USA) initiated a forest health monitoring program in 1990 to provide information about forest conditions in the USA. The program investigated four interrelated activities: detection monitoring, evaluation monitoring, research on monitoring techniques, and intensive site monitoring. The program's primary goal is to be aware of the changes in forests, to support decision-makers with accurate information, and to provide an annual review of the health status of forests [3].

The United States Department of Agriculture (USDA) created a program called Forest Inventory and Analysis (FIA) to monitor the status of the forest in the United States. The program depends on forest health indicators covering human needs within the acceptance budget. FIA defines several indicators such as crown condition, tree mortality and standing dead trees, tree damage, vegetation profile, and soil quality that can provide insights into forest health to forest managers [4].

To study and understand the status of the forest, forest managers usually need to harvest a large area to define the area's health status by exploring various ecological indicators depending on the goal of this process. This harvesting needs tools and experts to visit the field, requiring time and money. Developing a model to measure ecological indicators will reduce the costs of this issue. For example, Näsi et al. [5] compared the ability of UAVs to detect beetle damage against aircraft and found that UAVs can detect beetle damage better than aircraft in urban forests. Huo et al. [6] created a new index to detect the bark battle attack on European spruce forests to measure and detect tree damage.

Human activities, natural stressors, and disturbances can affect all levels of biological organization in forest ecosystems and their resilience. The relationships between factors, stressors, disruptions, and impacts are complex, often nonlinear, and multidimensional at the temporal and spatial levels [7]. Both species- and region-specific adaptive processes for forest species make it more challenging to understand causal stress responses and their effects on ecosystem resilience [8]. To understand forest health, scientists must investigate the different factors of stressors and disturbances to support forest managers and decision-makers in building their decisions upon a holistic approach. A holistic approach includes several elements, such as data recording, analysis, monitoring, and assessment of forest health [2].

The paper highlights different measurement approaches and their cooperative role in assessing forest health. Field observation, ground-based sensors, and remote sensing are key approaches used to measure various indicators that provide insights into the current health status of forests. Each of these tools has its own set of features and drawbacks. Field observations offer high accuracy and are considered one of the most reliable techniques for determining forest health status. However, they can be time-consuming and costly. For instance, Park et al. [9] identified the difficulty of accurately assessing tree attributes in tall, dense, multilayered forests.

Ground-based sensors provide valuable information regarding weather conditions and other indicators relevant to forest health. However, they are subject to limitations such as power shortages or network issues, affecting their continuous operation. Considering these limitations when relying on ground-based sensors for monitoring is essential.

Remote sensing enables measuring a wide range of forest health indicators at an acceptable cost. However, remote sensing also has its limitations, including the need for permissions, resolution limitations, and data quality variations. For instance, the resolution ability of satellite remote sensing in monitoring forest ecosystems sometimes needs improvement within the limits of feasibility [10].

Our review focuses on providing and identifying several aspects and techniques to study forest health. Our main contributions are the following:

- This review contributes by offering insights into forest health, forest index data, and innovative techniques for creating forest systems.
- We reviewed and identified several forest indexes and indicators by illustrating each part in a separate section and supporting it with examples. In addition, we presented the key considerations of a forest, the scales of the forest, and external factors that affect the forest.

- We explored various methodologies and techniques to address the challenges of forest health monitoring. The authors provided multiple methods for forest health monitoring, including field observation, ground-based sensors, and remote sensing. The authors identified several applications that have been used in forests. In addition, we illustrated the multiple data sources and provided published papers that used data fusion to enhance forest monitoring
- Lastly, we outlined the existing challenges and the future research direction of forest health.

1.2 Existing Surveys

Forest health has become an important topic to investigate regarding the impact of the forest on the ecosystem. In recent years, techniques and approaches have been developed to measure the sustainability of forests and provide information for forest management plans. However, they mainly focus on the measurement approaches to study forest health. This paper explores the forest health indicators and FHI affecting forests and defines several measurement approaches.

Prior survey articles have reviewed many of the techniques and approaches that we examined in this article. One review paper discussed implementing UAV-based on forest health, and the aim was to produce a review paper covering the requirements for developing forest health monitoring [11]. The authors reviewed 99 papers related to UAV-based forest health in the last 10 years. They identified the features and drawbacks of implementing UAVs in forests. As a result, the authors clarified the value that drones provide in enhancing the techniques of forest health monitoring. However, it is still hard to rely on UAVs alone to build a monitoring system for agriculture or forest.

Another review paper covered the criteria for developing forest health monitoring and discussed the methods of dealing with various data sources and presenting these data to create a decision-support system [7]. The article has covered several essential topics in developing a forest health monitoring system. In addition, it covered the technique of dealing with different data and identifying the difficulties or needs of ground-truth data for validation purposes to develop a monitoring system.

Further, Torres et al. [12] illustrated forest health issues using remote sensing techniques. The authors reviewed articles from 2015 to 2020 to show forest health issues using remote sensing techniques. They found several key features, such as the number of papers on this concern increased in recent years, the satellite method is one of the leading methods in this part, and most of the papers focused on two aspects: evaluating the impact of a specific stress or disturbance factor. By contrast, a few articles discussed the early warning technique. In addition, Camarretta et al. [13] studied the ability of remote sensing to study forest restoration, identify several remote sensing platforms, and investigate each forest structure attribute.

Another survey paper examined the importance of adding biodiversity as an indicator of forest management plans [14]. The authors of that paper selected 94 papers from the 1990s to 2020 that fulfilled these criteria: aspects of biodiversity (structure–composition–function) and four forest management categories (unmanaged, managed, plantation, and silvopastoral). In addition, they used three criteria to evaluate the papers: cost, time, and ease of operation on forest stand and landscape level. Lastly, Pause et al. [15] published a review paper that contained the value that satellite remote sensing provides in monitoring forest health and the role of field observation in enhancing the remote sensing technique for monitoring.

Most of the aforementioned papers explored the importance and ability of remote sensing in studying forests. However, this paper aims to analyze remote sensing techniques with other measurement approaches' techniques, such as field observation and ground-based sensors, intending to provide knowledge regarding the ability of these techniques to measure several indicators and the ability to use or develop an index. In addition, we highlight various parameters, including forest health and FHI, identify ecological and meteorological indicators for forest health, and discuss key considerations and factors that affect forest health. We distinguish our paper by providing an overview of forest health, index, and measurement approaches. Our primary objective is to give readers insights into the forest index and the development of forest monitoring systems. Table 1 compares the approaches of different review papers. In this table, we present a comparison of various review papers, shedding light on their respective approaches to understanding the concept of forest health.

Moreover, the table presents the number of papers considered and analyzed to provide valuable insights. Further, we highlight if a paper covered the definition of forest health or if the authors clarified their paper's analysis or evaluation techniques. Lastly, the researchers described the multi-data sources approach in their article "Integration."

1.3 Structure of the Paper

The obtained papers were categorized based on their relationship with the FHI and the data-gathering techniques. Section 2, covers the methodology of collecting documents. Section 3, provides fundamentals of forest health by

Author	App	roach		Year	Studies		Forest Heal	th		
	RS	SO	GB		Considered	Analyzed	Definition	Methods	Evaluation	Integration
[15]	\checkmark	\checkmark		2016				\checkmark	\checkmark	\checkmark
[7]	\checkmark			2018				\checkmark	\checkmark	
[13]	\checkmark			2020					\checkmark	\checkmark
[14]				2020	188	94		\checkmark	\checkmark	
[12]	\checkmark			2021	3722	107		\checkmark	\checkmark	
[11]	\checkmark			2022	1073	99		\checkmark	\checkmark	
This paper	\checkmark	\checkmark	\checkmark	2023	210	90	\checkmark	\checkmark	\checkmark	\checkmark

Table 1: Previous survey comparison. Our uniqueness is to study the three measurement approaches intending to generate a forest index

RS: Remote Sensing. **SO:** Site Observation. **GB:** Ground-Based Sensor.

defining several elements that affect forests, including key considerations of forest, indicators, indexes, scales, and external factors. Section 4 covers the measurement approaches of forest health, such as field observation, ground-based sensors, and remote sensing. Section 5, covers multi-data sources. Section 6 covers the findings in this paper. Section 7 contains research challenges and directions—and finally, the conclusion section.

2 METHODOLOGY

This section discusses the survey method used in this work. To generate this review, the authors focused on the papers related to studying forest health index. This paper focuses on more than just the measurement approach or tool. Still, it discusses several indicators, indexes, and measurement approaches to provide a better vision of forest health monitoring and forest index, which could lead to understanding forest health.

2.1 Research Questions

This review aims to provide insights into forest health, forest index, and measurement approaches that help enhance the ability to monitor the forest health status. Accordingly, the study focuses on answering the following questions.

RQ 1. What is a forest health index? : Indexes offer valuable information regarding specific aspects of a forest, for example, the normalized difference vegetation index (NDVI) and leaf area index (LAI). The NDVI quantifies the presence of green vegetation in a defined area. Consequently, no definitive directive regarding creating or utilizing an existence index exists. Instead, researchers typically select or create an index based on the objectives of their study, such as an FHI.

RQ 2. What indicators are there for studying forest health? : Exploring various indicators becomes paramount in the pursuit of understanding and safeguarding the health of our forests and ecosystems. These indicators span different dimensions, encompassing biological, meteorological, and ecological factors. For instance, delving into meteorological indicators like temperature patterns can revolutionize forest health monitoring. We can detect potential fire outbreaks in their early stages by establishing a robust forest health warning system based on temperature fluctuations. Moreover, using multiple indicators promises a more comprehensive and accurate understanding of a forest's well-being. This research explores the probability of combining these diverse indicators to enhance our forest monitoring capabilities, ultimately contributing to the preservation and sustainability of these vital ecosystems.

RQ 3. What are the techniques for studying forest health? : In forest ecology, the choice of measurement techniques is pivotal, with options ranging from remote sensing to ground-based sensors and traditional field observations. Our research determines several techniques used to measure forest health. For example, the advent of drone and satellite technology has substantially widened our horizons, enabling us to monitor expansive forested regions with unprecedented precision. Our study delves into these questions, aiming to provide valuable insights regarding measurement techniques for forest indicators, ultimately advancing our understanding and conservation of these vital ecosystems.

RQ 4. How do multi-data sources help in monitoring forest health? : The multi-data sources approach in forest health monitoring considers using different types of sensors to gather data regarding forest objects and the ecosystem. Multi-sensor advantages include enhanced data accuracy, improved spatial coverage, and the ability to capture diverse aspects of forest health.

Section 3 will answer the first two questions, Section 4 elaborates to answer question 3, and Section 5 covers question 4.

2.2 Research Methods

To conduct this review, the authors searched several digital libraries, including Google Scholar, Web of Science, IEEE Explore, ACM Computing Survey, and SpringerLink, using the following keywords and phrases: "Forest Health Index," "Physical Environmental Indicators," "Forest Health Monitoring," "Remote Sensing," "UAV," "Tree Health," "Field Observation," "Site Observation," "Ground-Based," "Temperature," "Precipitation," "Stream Flow," "Snowpack," "Ozone," "Soil Moisture," "Multi-Sensor Data Fusion," and "Site Measurement." This review used the Boolean operators "AND" or "OR" to enhance the results. The study's article search and selection method was divided into different stages. The first stage was the search for forest health monitoring, indexes, and indicators. The second stage was the search for remote sensing techniques and approaches for forest health monitoring. During this phase, we examined UAVs more regarding several features that drones provide. The third stage was the investigation of the field observation tools and techniques. The fourth stage was the exploration of the different ground-based sensors—finally, the multi-model (multi-data sources) measurement approaches, especially data fusion.

A total of 210 research papers regarding forest health monitoring, FHI, and forest health measurement approaches were reviewed. These papers were filtered by removing duplicate papers or focusing more on a wildfire or wildlife, reading their title, abstract, and conclusion, and then filtered by full-text reading. This filtration resulted in 90 articles Figure 1.



Figure 1: Survey methodology overview: Review Process Illustration.

3 FOREST HEALTH

This section presents an overview of forest health, including the critical considerations of studying forest health (in section 3.1); the authors illustrate the three key considerations of studying forest health (Evaluate the current state, Study the area's history to generate a future prediction system and Generate a warning system). Indicators (in section 3.2), we discussed the three indicators (Ecological, Meteorological, and Biological) as each feeds the idea of forest health from different perspectives. Indexes (in section 3.3): In that particular section, we explore the two most popular indexes (NDVI, LAI) and supply them with determining the published papers focusing on forest health and containing the FHI in the paper's title. Scale (section 3.4) describes the importance of scaling in studying forest health and identifies the three scales (Single Tree, Stand scale, and Landscape-scale). External factors (in section 3.5) define the other factors impacting forest health, such as natural phenomena.

3.1 Forest Key Considerations

There are various considerations for studying forest health. One crucial consideration of forest research involves evaluating the current state of the forest, a practice often referred to as evaluation monitoring. This approach allows researchers to assess various factors influencing forest ecosystems. For instance, in a study conducted by Frey et al. [16], the impact of microclimate and vegetation levels on bird distribution in mountain landscapes in Oregon, USA, was thoroughly investigated as the distribution of birds will help the bioscientist directly or undirect to have a sign about the situation level on that area.

Evaluation monitoring plays a pivotal role in gaining insights into the health and dynamics of forests, helping researchers and conservationists make informed decisions for preserving and managing these areas. It involves an examination of ecological, climatic, and biological variables, contributing to a deeper understanding of the intricate relationships within forest environments.

Another valuable consideration of forest research involves delving into forested regions' historical trends and changes. This exploration can encompass various parameters, including shifts in vegetation levels and alterations in water availability. Understanding the historical context is instrumental in predicting future conditions in these areas, a task often facilitated by applying machine learning models such as random forest.

For example, Roshani et al. [17] employed the random forest technique to assess temporal changes in India's environmental conditions. By analyzing data from the Indian Meteorological Department from 1981 to 2020, their study not only shed light on past transformations but also offered insights crucial for anticipating future developments. Investigating the historical record of forests is a fundamental building block to ensure their continued health and resilience.

The last consideration of forest research encompasses exploring warning systems, a focal component in forest management and conservation. These systems can be categorized into two fundamental types: first, the ability to trigger an alarm when an unusual event occurs; second, the capacity to issue advance warnings based on predictive models. The prediction of advanced warnings is greatly facilitated by applying statistical analysis techniques and machine learning methodologies. For example, Chen et al. [18] harnessed the power of machine learning in the pursuit of precise wildfire prediction. They leveraged a comprehensive dataset containing RGB and thermal fire images, demonstrating how cutting-edge technology can significantly enhance our ability to forecast and mitigate forest-related disasters. Understanding and refining warning systems in forest ecosystems not only aids in safeguarding these invaluable natural resources but also holds the potential to reduce the impact of catastrophic events, thereby contributing to the sustainable management of our forests for generations to come.

3.2 Forest Indicators

Ecological, meteorological, and biological indicators are elements that aid in studying forests and provide insights to determine the health status of the woods. Figure 2 outlines the envisioned system structure of this review paper, depicting the various factors that influence forest health and identifying the appropriate measurement approaches for forestry. In this part, we will describe the indicators in detail; later, in section 4, we will explain the measurement approaches. The first indicator is ecological indicators in the context of forests, which are measurable characteristics or variables that provide insights into an ecosystem's overall health, functioning, and dynamics. These indicators assess forest ecosystems' ecological condition, diversity, and sustainability. They help researchers, ecologists, and land managers monitor and understand how ecosystems respond to environmental changes, disturbances, management practices, and human impacts. Several indicators can be used to evaluate the environmental condition of a forest, including vitality, productivity, and biodiversity. Vitality refers to the overall health and vigor of the forest. The quality of tree attributes such as good growth rates, crown condition, and tree damage mainly represent it. Productivity is the ability to produce resources such as timber, tree diameter, and tree height, which are crucial in measuring productivity [19]. Biodiversity refers to the variety and abundance of species within a forest ecosystem [20]. For example, Arwanda & Safe'i claimed that vitality, productivity, site quality, and biodiversity would help determine a forest's health status in Indonesia. The results showed that cluster plots 1, 2, and 4 were in good condition, and 3 were in bad condition. Furthermore, Yang et al. [21] studied the national monitoring program in China to evaluate the ecological function and compared it with other international programs.



Figure 2: Forest health measurements approaches and indicators. The figure represents different factors that affect the forest status and the techniques to measure these factors.

Second **meteorological indicators**, such as temperature, precipitation, and snowpack also affect the monitoring system. Meteorological indicators in this context refer to specific weather-related variables and measurements that play a crucial role in understanding the environmental conditions within forest ecosystems. Using these indicators,

researchers, ecologists, and forest managers may evaluate the effects of meteorological conditions on the health, development, and dynamics of forests. Meteorological indicators are essential for studying various ecological processes and making informed decisions regarding forest management, conservation, and climate change adaptation. For instance, temperature significantly affects the development of trees, which are crucial indicators for describing the health status of the forest. Song et al. [22] explored the impacts of air temperature, tree species, and leaf size on the tree surface temperature in tropical forests. They found that various tree species have different leaf and air temperature differences. In addition, a tree's temperature changes due to climate change depend on the size of its leaves and the amount of air that it releases through tiny pores (stomata) on its surface. Trees with small leaves and high air release are less affected by climate change. Furthermore, Marsh et al. [23] generated a model that studies the correlation between forest structure and air temperature. The result showed that nearly all structural variables significantly differed between vegetation plots, and they recorded a wide range of variation as they found variation by 15.2 °C between data loggers. Moreover, there was a variation of 14.8 °C between data loggers simultaneously at different heights in the same tree. Other meteorological indicators are precipitation and snowpack, which are essential for assessing water availability in a given area [24].

The third **biological indicator** is living organisms or biological parameters used to assess the health of the forest. Several researchers studied the impact of air temperature on the birds' distribution and breeding phenology. For instance, Shutt et al. [25] investigated the effect of microclimate air temperature on three different types of birds in a UK forest. They found a connection between the microclimate air temperature and breeding phenology. In addition, the paper investigated the spatial variance and measured several factors to understand the breeding phenology for all three birds. Han et al. [26] studied the breeding habitats for black-necked cranes in Central Asia using ML. The result showed the ability to predict the birds' breeding distribution using the species distribution model. Furthermore, they presented the rank of each factor that affects the bird. Iijima et al. [27] examined the dynamic seasonal change in bird assemblages. Biological indicators offer essential information regarding the general condition of an ecosystem and the effects of environmental changes or disturbances in the context of environmental monitoring or ecological evaluations. In Table 2, the authors summarized several papers using the three indicator types. Also, the table illustrates the different objectives for each and identifies various notes in each article.

Some other indicators or factors can be highlighted in this paper as they provide different points of view in studying forests. These factors are elevation and distance from the road. Elevation often affects air temperature, as a higher elevation will generally experience a cooler temperature. This factor can help in understanding the habitat preferences of different species. For example, Wu et al. [28] studied the impact of elevation and the relation of temperature and humidity on the growth of leaf phenology in three plant types in a subtropical forest in China. Huerta et al. [29] studied snow depth at different elevations and locations within the forest. The distance from the road is another factor that highlights the impact of humans on forests. At the same time, Han et al. [26] considered the distance from the road as a factor when studying the breeding habitats for the black-necked crane. Mi et al. [30] used the distance from the road to generate the best possible prediction habitat for great bustards. All of these factors can be classified into various categories, such as physical characteristics or human impact.

Table 2: Summary of indicators

Paper	Indicators		ors	Objective	-
	Е	М	В		
[19]	0			The paper aims to assess the health status of Panca Indah Lestari Community Plantation Forest, a plantation forest.	_
[20]	0			The paper aims to study the health status of a Conservation forest in Indonesia.	
[22]		0		The paper aims to study the impact of meteorological events, especially air temperature, on plants.	E
[23]		0		The authors are studying climate change in Indonesia's degraded tropical forest.	
[24]		0		The paper aims to assess the bulk snow isotopic in forested (pine and birch) and open areas.	
[25]			0	The paper aims to investigate the impact of microclimate air temperature on three different types of birds in a specific UK forest.	
[26]			0	The paper aims to study the breeding habitats for Black-necked Cranes in Central Asia by using ML.	
[27]	•	•	0	The paper aims to study the dynamic seasonal change in bird assemblages in a specific mountain in Japan.	

Ecological. M: Meteorological. B: Biological. ○: Refers to the main focus of the paper. ●: Use as a support factors.

3.3 Forest Indexes

Scientists consider several indicators to study forest health by utilizing or creating indexes. Spectral indexes, such as the NDVI and LAI, are commonly used to assess forest health. These indexes provide valuable information regarding vegetation cover, health, and productivity, which are essential for forest ecosystems. The NDVI refers to the ability to measure the amount of green vegetation in a given area, and it is calculated based on the reflectance of near-infrared (NIR) and red light wavelengths captured by remote sensing. In remote sensing, bands refer to specific wavelength ranges or channels in the electromagnetic spectrum captured by sensors or cameras. These bands recorded data and captured information regarding the Earth's surface or the observed objects. Landsat measured NDVI differences to

assess the forest's change to determine the forest's health status in Italy [31]. Bolten et al. [32] used the NDVI with other factors, such as soil texture, to monitor crop growth stage and condition and, subsequently, globally forecast agricultural yields. However, we focused on identifying all forest indexes, not just spectral ones.

The LAI is another popular index in studying forestry, agriculture, climate change, and biodiversity. The LAI is defined as the ground area covered by the plants [33]. The LAI is an essential factor in studying forest health. Although in situ measurement is one possible technique to measure the LAI, it is time-consuming. Conversely, a remote sensing technique can provide a solution to measuring the LAI with an acceptable accuracy rate. Pope & Treitz [34] showed the importance of LiDAR in measuring the LAI. By contrast, both the NDVI and LAI are unsuitable for directly measuring individual trees' physical attributes.

We investigated several articles containing the "forest health index" between 2000 and 2022. We found a few papers that have directly used the word FHI in their title. For example, Olthof & King [35] created an FHI (image-based health index) regarding the most significant image spectral, textural, and radiometric fraction measures. Huo et al. [6] developed a new index, Normalized Projected Red & SWIR (NRPS), to detect the bark beetle attack on European spruce forests. The NRPS index used a red band and a shortwave infrared (SWIF) band to detect the bark battle on a tree. Winarso et al. [36] proposed a new index aimed at developing a satellite-based Mangrove Index as an alternative to NDVI. Unlike NDVI, which relies solely on one parameter, this novel index offers a more comprehensive approach to monitoring mangrove forest health.

Most of the aforementioned papers used remote sensing to develop indexes, and they measured different scale sizes. Table 3 summarizes selected papers focusing on the FHI.

	Table 3: Summary FHI							
Paper	Index	Aspect	Time-Frame	Sensor & Dataset Information				
[35]	Image-based health index	Tree growth	Mid of Augests 1997	Multispectral				
[36]	Satellite-based mangrove index	Tree growth	April 10, 2018	Landsat 8				
[37]		Tree	July 27, 2009,	Hyperspectral				
[38]	Floristic quality assessment index (FQAI)	Species diversity	2018-2021	Existence dataset- survey collection				
[6]	Normalized projected red and SWIR (NPRS)	Bark beetle (insect-disease)	04/07, 07/26 and 10/07 of 2019	Sentinel-2 satellite				

Table 4 shows several indexes that can be used for several aspects. This table identifies several indexes and shows that the primary index used is the NDVI, and the second is the LAI. The table exposes several elements. Indexes can help study several aspects, for example, using the NDVI as a supporter factor to investigate the tree species or the probability of fire predictions. At the same time, the Aspen Center for Environmental Studies has launched a website dedicated to exploring and monitoring Colorado State's forests. Recognizing these forests' critical role in the local ecosystem, the center's mission is to aid in future planning and to safeguard this vital ecosystem component. To achieve this, they have developed the FHI, a tool that annually measures 12 key indicators on a large scale. These indicators are temperature, extreme temperature, precipitation, frost-free days, stream flow, bear mortality, snowpack, soil moisture, critical fire risk, resource use, ozone levels, and insect and disease prevalence. This website is a valuable resource for understanding and preserving the health of Colorado's forests and the broader ecosystem they support [39].

In conclusion, forest health assessment involves a complex interplay of various factors such as the forest type, geographical location, elevation relative to sea level, distance to roads, and prevailing climatic conditions. The diverse nature of these elements within each forest presents a challenge when establishing a specific index for assessing or monitoring a particular area's health. Developing such an FHI demands effort and careful consideration. In addition to the factors mentioned earlier, it is essential to emphasize the selection of appropriate indicators and tools.

3.4 Scales

Scaling is essential in studying environmental systems and a forest's ecosystem. Understanding forest health requires considering various scales, such as stand, landscape, and individual tree levels. Each scale offers unique insights into ecological processes, impacts, and management strategies. Scales refer to different analyses and perspective levels of forest health. For instance, a single tree refers to individual trees, a stand scale refers to a defined area, and a landscape scale refers to a broader area than a stand scale. In addition, other words can be used to describe the size or the scale, such as the fine scale, which mainly covers the same area as the stand scale. Several researchers identified directly or indirectly the scale's effect in developing the monitoring system model. For example, Ćosović et al. [14] identified the remote sensing technique for data collection as more cost-effective and faster than the field at both stand and landscape scales. Moreover, studying a tree as a validation indicator in different forest areas (production forest, protection forest, and conservation forest) at the landscape level plays a role in understanding the forest's status [57].

Table 4:	Summary	of the	indexes.
----------	---------	--------	----------

Paper	Index														Aspect	Sensor Type
	NDVI	LAI	VI	NDWI	SMI	SDI	TVDI	FWI	GNDVI	NDSI	RSI	DSI	R.A.	CVI		
[40]	~		\checkmark	\checkmark										\checkmark	Crop Management- Agriculture	Satellite
[32]	\checkmark														Agriculture	Satellite
[41]	\checkmark						\checkmark								Soil Moisture	Satellite
[42]	\checkmark														Soil Moisture	Satellite
[43]	\checkmark														Forestry	UAV
[44]	\checkmark														Fire Detection	UAV
[45]	\checkmark														Canopy Fuels	UAV
[46]	\checkmark														Above Ground Biomass	UAV
[47]	\checkmark														Forest Tree Phenotype	UAV
[48]	\checkmark								\checkmark						Herbicides	UAV
[49]	\checkmark								\checkmark	\checkmark	\checkmark	\checkmark	\checkmark		Tree Defoliation	UAV
[50]	\checkmark	\checkmark													Tree Height	UAV
[51]	\checkmark	\checkmark													Generate Forest Maps	Satellite
[52]		~				~									Tree Damage	Satellite
[53]		\checkmark													Forest Health	Satellite
[54]		\checkmark													Tree Volume	UAV
[55]		\checkmark													Leaf	UAV
[56]					\checkmark										Crop Management- Agriculture	Satellite
[57]						\checkmark									Tree Status	Satellite
[58]								\checkmark							Fire Detection	

UAV: Unmanned aerial vehicle.NDVI: Normalized difference vegetation index. LAI: Leaf area index. VI: Vegetation index. NDWI: Normalized difference water index. SMI: Soil moisture index. SDI: Stand damage index. TVDI: Temperature vegetation dryness index. FWI: Fire weather index. GNDVI: Green normalized difference vegetation index. NDSI: Normalization of the different spectral indices. RSI: Spectral ratio index. DSI: Differential spectral index. R.A.: Reflectance absorption index. It summarizes the indexes in the remote sensing approach section 4.3. CVI: Cumulative vegetation index.

3.5 External Factors

Other factors must be highlighted in studying forest health, including natural disasters such as wildfires or floods, which are hard to prevent. Still, scientists can provide solutions to minimize the loss due to these events. Researchers have developed techniques to reduce the impact of natural disasters, or even prevent them when possible, by utilizing various indicators. These techniques include building prediction models to forecast fires and developing systems to detect fires in their early stages. For instance, one approach involves utilizing a drone equipped with a thermal camera to detect wildfires [18], thereby minimizing forest damage. In addition, studies have been conducted to assess the capabilities of UAVs in measuring canopy fuels and forest structures in the United States to mitigate wildfires in the region [45].

Flood is another natural phenomenon that can affect a city and a forest. Floods can significantly affect forests, disrupting the delicate balance of these ecosystems. Accordingly, Kim et al. [59] generated a multi-sensor system to explore the ability to examine the elements behind flood severity in Cambodia. The authors investigated the relationship between deforestation and flooding in Cambodia, the ninth most vulnerable country to natural disasters in the world, in 2011.

4 MEASUREMENT APPROACHES

Forest health status monitoring has gained significant attention in recent years in public discourse. Scientists and researchers have been actively exploring various methodologies and techniques to address the challenges of forest health monitoring after defining forest health and the elements and attributes used to measure forest health in Section 3; this section will cover three different approaches and tools to measure forest health. The first is field observations in section 4.1, then ground-based sensors in section 4.2, and the last is remote sensing in section 4.3.

4.1 Field Observations

Field or site observation is a method to investigate specific phenomena in a particular location to measure a certain aspect. It requires a physical presence on the site to gather data. Field observation plays a crucial role in forest health monitoring as it provides the ability to assess the ecosystem of forests, allows accurate data collection, and aids in developing a support decision system. The field observation or site observation in this review refers to researchers, scientists, ecologists, or naturalists investigating the forest health status from certain aspects to provide information or measure specific elements.

Data gathering for field or site observation usually requires specific tools and a certain amount of knowledge. This technique involves being physically present in the field to study various aspects of the environment, including ecological or biological indicators. Tools such as meters, cameras, tapes, binoculars, and sheets are examples of what a researcher can carry during the investigation. For instance, binoculars are one of the tools that researchers usually carry during their field observation to investigate different elements, such as the existence of birds, birds' diversity, tree species, and the

tree level of damage. Binoculars are used to explore bird diversity in Sabah, Malaysia, to determine the health condition of the forest [60]. Furthermore, Doria et al. [61] claimed that studying primates' existence indicates identifying this area's health status. During observing time, researchers may perform techniques such as visual observations, data collection through sampling methods, recording measurements, taking photographs, or making sketches.

Human observations and laboratory work will help investigate the impact of pests and diseases on plantation forest health in Vietnam [62]. Tree crown condition and diversity can be measured using a 50-m magic card, camera, note-taking pad, stationery, and tally sheet [63]. Other tools can be used during field observations, such as a digital laser tape measure to measure tree height or a camera to document the existence of birds or investigate tree diversity.

Figure 3 illustrates several tools that researchers can use, one or all, during the site observation. Also, we identified several attributes that can be measured using these tools. The format refers to the techniques of cleaning the data to make it more beneficial. Then, the authors define two analysis techniques as an example to process the data and provide an idea about the health status of the specific area.



Figure 3: Overview of field observation system structure.

Field observation involves collecting and analyzing data to gain insights into ecological processes to validate existing theories or hypotheses, species interactions, and environmental changes. There are several techniques to study field observation data, such as descriptive statistics, spatial analysis, data mining, and machine learning. These techniques can help generate a system to measure a particular area in a forest. For example, Pranolo & Widyastuti [64] developed an intelligent agent for urban forest health monitoring using the simple additive weighting (SAW) method. The authors used the iLIS software to represent the collected data to the users and used SAW to analyze the data. As a result, the authors help the end users document their work and understand the health status of a particular area from a certain point. In addition, the machine learning approach played a role in developing a model (boosted regression tree models) to study the variables that influence the vegetation plots and the air temperature [23].

Field observation can provide accurate information regarding a specific phenomenon or situation and is used as groundtruth data to validate the results from other approaches. In addition, it can measure different ecological or biological indicators essential to forest health management. By contrast, the drawbacks of this method are high cost and limited accessibility, which make scientists investigate other techniques to evaluate, predict, and monitor the health condition of a forest. One of these techniques is remote sensing, which can streamline data collection, allowing more information to be assessed more accurately and efficiently [14]. However, further research must ensure that these tools are accurate enough to be reliably used in various ecologies across different geographical scales. Until then, researchers can rely on data collected from physical surveys of forest stands (looking at factors such as tree size and density) to help inform management plans and decision-makers and to prepare forests for an uncertain future.

4.2 Ground-Based Sensors

Ground-based sensors refer to sensors installed in a forest to gather data regarding the environment of this plot. Groundbased sensors play an essential role in determining a forest's health condition. These sensors measure temperature, soil, humidity, snowpack, and vibration. Soil moisture is identified by FIA as one of the indicators for forest health monitoring in the United States. In addition, temperature dramatically affects tree growth, vegetation level, and tree crown condition. Regarding ground-based sensors, we overview the system structure in Figure 4. A network of diverse sensors is strategically positioned in the field to continuously monitor real-time data, encompassing variables like temperature, humidity, and soil moisture. The gathered data seamlessly transfers to the nearest computing or edge node through wireless communication protocols such as WiFi, LoRaWAN, and NB-IoT. Subsequently, the edge node undertakes the crucial task of pre-processing the data before promptly transmitting it to the designated server in the cloud via the Internet gateway. Within the cloud infrastructure, one or more servers serve as a database and data processing hub, efficiently handling all collected data from the field. Concurrently, this setup facilitates the visualization of post-processed data on a user-friendly dashboard.



Figure 4: Overview of ground-based sensors system structure.

This section investigates and targets different indicators that are related to ground-based sensors. Those indicators, like temperature, snowpack, stream flow, precipitation, ozone, soil moisture, and fire risk, are identified by the Aspen Centre for Environmental Studies. The center studies the FHI in Colorado. The center identified 12 indicators that provide information regarding the health status of the forest to the forest management department on their website [39]. In addition, it influences other countries and societies to develop a forest health monitoring system, especially in places that are affected by wildfires. This section targets different meteorological indicators and other indicators that are related to ground-based sensors. Temperature is an essential parameter in meteorology and is commonly measured using various types of temperature sensors. Choosing the type and style of the sensors relates to the project's objective, the techniques for collecting data, and the budgetary cost of the project; all of these factors play a role in determining the type of sensors. By collecting temperature data, we can have insights into the health condition in that area by applying different types of analysis. The authors generated a Sankey diagram for ground-based sensors with different indicators. The summary is shown in Figure 5. The first level is the approaches, followed by the second level for sensor types, the third level for the applied model, the fourth for the domain, the fifth for the research subject, and the forest health.



Figure 5: Sankey diagram for ground-based sensors with different indicators summary.

• **Temperature** plays a role in tree leaf phonology, as Wu et al. [28] studied the impact of elevation and the relation of temperature and humidity on the growth of leaf phonology in three plant types in a subtropical forest Chain. Xu et al. [65] used air temperature as a factor to generate a model to predict leaf growth. Studying temperature leads to understanding ecosystems; different forests react differently to weather. Wiesner et al. [66] investigated the ecosystem's response to extreme heat and cold events in three long leaves in the savanna to explore the impact of climate on forest health. Also, temperature plays a role in birds' breeding phonology as it is affected by the temperature as Shutt et al. [25] provided a research paper regarding territory-level temperature influences on breeding phonology and reproductive output in three forest pas-serine birds. In addition, several researchers have approved the impact of temperature on bird distribution [16] and [67] bird abundance.

As we discussed above, the impact of temperature sensors in understanding the forest, here we illustrate several types of sensors: ATMOS-14, Ther-machron iButton DS1921G, HOBO UA-002–08 8 K Pendant Temperature, SM2110, and SHT31 sensors. In Table 5, we will summarize several papers covering different sensors in the ground-based. Moreover, they illustrate several indicators and define the models, aspects, and study areas.

Paper	Indi	cator	rs			Model	Aspect	Location	•
	Т	Р	S	SM	FR				
[28]	\checkmark					LR	Tree leaf growth	Chain	
[65]	\checkmark					AAT	Tree leaf growth	China	
[68]	\checkmark					MG	Tree leaf growth	USA	
[25]	\checkmark					LM	Breeding phonology	U.K.	
[16]	\checkmark					DO	Birds distribution	USA	
[67]	\checkmark					N-mixture - PCA	Bird abundance	USA	
[23]	\checkmark					ML- BRT	Forest structure	Indonesia	
[69]	\checkmark						Forest ecosystem	China	
[70]	\checkmark					RF	Water stress index	China	
[66]	\checkmark						Agriculture	USA	
[22]	\checkmark					LR	Plant productivity	China	
[71]	\checkmark	\checkmark					Cycling of precipitation	Switzerland	
[24]			\checkmark			MLR	Water isotope	Siberian	Т:
[72]			\checkmark			Semivariogram	Scaling	Canada	
[73]			\checkmark			SNOW-17	Watersheds	USA	
[74]			\checkmark				Hydrological, meteorological	Canada	
[75]			\checkmark			CLS	Cold content	Canada	
[29]			\checkmark			MLR - LPL	Interaction between vegetation and snow processes	Chile	
[76]			\checkmark			LM	Bark beetle		
[77]				\checkmark		PC	Volumetric soil water content	Malaysia	
[78]				\checkmark		TPHT	Soil respiration	India	
[79]				\checkmark		RMSD- CC	Soil moisture	Mexico	
[80]				\checkmark		LM	Depth to water	Canada	
[81]					\checkmark	MLR	Fire risk	Brazil	
[82]					\checkmark	ML	Fire risk	Brazil	
[83]					\checkmark	ML	Fire risk	Thailand	
[84]					\checkmark	ML	Fire risk	Algeria	
[85]					\checkmark	ML	Fire risk	Lebanon	

Table 5: Summary of several indicators for ground-based sensors

Temperature. P: precipitation. S: Snowpack. SM: Soil Moisture. FR: Fire Risk. "LR": Linear regression. "LM": Linear mixed. "ML": Machine learning. "BRT": Boosted regression trees. "RF": Random forest. "AAT": Accumulated temperature. "PCA": Principal component analysis. "MLR": Multiple regression. "RMSD": Root Mean Square Deviation. "Micrometeorological": MG. "DO": Dynamic Occupancy. "CLS": Canadian land surface scheme. "LPL":local polynomial. "PC": Pearson correlation. "TPHT": ANOVA-Tukey post hoc test. "CC": Correlation coefficient

• **Snowpack** refers to the amount of snow on a mountain during winter. This meteorology indicator provides an angle for understanding the expected amount of water in a specific place. This indicator can be detected using different approaches, but we focus here on ground-based sensors. Studying the snow's water isotopes will help trace hydrological and ecological processes. Exploring the impact of the forest canopy in snow bulk can help understand hydrological and environmental processes in forested (pine and birch) and open areas [24]. Moreover, scaling and location affect snowpack melting. Beaton et al. [72] investigated scaling issues by measuring the snowpack in a northern Great Lakes-St Lawrence forest. Another aspect that influences the snowpack is the type of snowpack, as they are a factor in investigating the amount of net water input to the soil [73]. Various ground-based sensors are there to measure the snowpack, such as a 60 cm snow coring sampler VS-43, SR50, and an ultrasonic snow depth sensor.

- Soil moisture's stability is an essential indicator of forest health and is considered a fundamental data source for agriculture. In addition, soil moisture is critical in the study of climate change [86]. Soil moisture refers to the amount of water being held in the ground. Remote sensing and ground-based sensors are the most common techniques for measuring soil moisture. There are several published articles regarding remote sensing techniques measuring soil moisture from different perspectives, for example, [87], [88], and [32]. Another technique is a ground-based sensor where [77] used PR2 sensors to investigate the spatial distribution of volumetric soil water content (VSWC) in tropical rainforests. Further, many soil moisture sensors are available to measure soil moisture, including PR2, ADR, and Q-Box SR1LP.
- Fire risk is another indicator for forest health monitoring. (The European Forest Fire Information System) saw an average of 1 million acres burned annually between 2010 and 2019, including countries from the Middle East and Northern Africa (CDP, 2022). As a result of the wildfire, Delgado et al. [81] created a new forest fire index in Brazil to reduce forest damage. In addition, Dubey et al. [89] explored the ability to detect fire in an early stage to reduce the amount of damage. Moreover, Kelleher et al. [90] developed a low-cost system to evaluate PM2.5 to study the forest fire risk. In line with Lertsinsrubtavee et al. [83], the researchers employed a cost-effective wireless sensor network to identify forest fire incidents in Thailand, focusing on the PM2.5 and CO parameters. They utilized the J48 classification algorithm and introduced a decision tree model to predict the risk of forest fires. Additionally, other studies [84] [82] also adopted the J48 classification model, incorporating meteorological variables such as temperature, relative humidity, and wind speed to detect forest fires in Algeria and Brazil, respectively. In a distinct approach, Karouni et al. [85] utilized the ID3 algorithm for forest fire detection in Lebanon, relying solely on temperature and relative humidity.
- Ozone refers to a chemical (O3) compound found on the upper level of the atmosphere. Ozone affects different attributes of a forest as it directly affects tree growth. Investigating the effect of temperature and ozone on tree health can give insights into the impact of ozone on forest health [91].

4.3 Remote Sensing

Over the past decade, remote sensing has revolutionized the ability to monitor forests with an acceptable accuracy rate. With different accuracy capabilities, remote sensing can measure several forest attributes, such as the vege Remote level and tree crown. Remote sensing has different definitions: for example, Mcroberts et al., 2010 [51] defined it as observing and sensing the Earth's surface from a distance. In addition, Nicholas M. Short [92] described it as detecting and measuring radiation, particles, and fields from things beyond the location of the sensor device. Another definition of remote sensing is science, art, tool, or technique [93]. Satellites, manned aircraft, or near-surface (drone) sensors are tools utilized in remote sensing. Figure 6 illustrates the three measurement approaches, the relation between operations, the spatial resolutions, and the different scales.



Figure 6: Different approaches of remote sensing.

4.3.1 A satellite

is an object intentionally placed into orbit for several purposes, including communication, weather monitoring, and scientific research. A satellite in this review relates to collecting information regarding forests from different perspectives, such as vegetation level, environmental change, or other goals. Satellite usage dramatically affects studying various

forest parts, including forest health monitoring, generating a warning system, and developing a prediction model. We illustrate several published papers demonstrating the use of several satellite applications that have contributed to various forest aspects. For example, for crop management and agriculture, Becker-Reshef et al. [40] discussed the GLAM monitoring system's operational components and new developments and the future role of Earth observations in global agricultural monitoring, especially in studying timely food supply information.

Mohamed et al. [56] studied the relationship between the amount of soil moisture in the ground and crops in Egypt, specifically the Nile Delta, based on remote sensing data and synthetic aperture radar (SAR) Sentinel. In addition, Chen et al. [41] and Ahmad et al. [42] investigated soil moisture using satellite remote sensing. Chen et al. studied if the temperature vegetation dryness index is suitable for estimating soil moisture and if soil moisture is significantly affected by tree species in the Laoshan forest. Ahmad et al. studied the ability to evaluate soil moisture content using remote sensing data for the selected Lower Colorado River Basin sites.

Lu et al. [53] focused on five forest farms in Beijing, identified the influencing factors of forest health, and examined how they contribute to revitalizing rural areas. The research paper explored the pattern of health conditions for different types of forests, based on their age and category, such as young, middle-aged, near-mature, mature, and over-mature forests. In addition, it looks at the conditions in the "shelter forests" and "special-purpose forests" categories. Anwar et al. [57] aimed to understand the characteristics of three types of forests by identifying landscape characteristics regarding the levels of damage to particular kinds of trees in three different forest functions. Tian et al. [94] explored how different sensors, such as satellite and aerial stereo camera systems, can monitor and detect changes in forest areas. Moreover, satellites can detect and monitor insect attacks and catch fire. For example, Sahin [52] observed the larch forest insect in the early stage, and Wang et al. [58] proposed the used of classified animal tracking data and thermal data for forest fire detection, using animals as mobile biological sensors(MBS). Stojanova et al. [95] leveraged datasets representing various regions of Slovenia, namely Kras, Primorska, and Continental Slovenia. Their approach encompassed the utilization of several variables for the purpose of forest fire detection, incorporating a comprehensive array of data sources such as geographic information systems (GIS), MODIS imagery, and meteorological data.

There are several features and drawbacks to satellite monitoring a forest. One feature that satellites provide is the ability to monitor a wide range of areas. However, one of the drawbacks of a satellite presented by Ecke et al. [11] is that satellite imagery, used successfully in temperate and boreal regions to record phenological patterns and their changes in response to climate, is more difficult to interpret in tropical forests. In addition, weather conditions and cloud cover make it hard to collect continuous time series data for multispectral imagery.

4.3.2 Manned aircraft

or human-crewed aircraft is a remote sensing technique requiring a pilot. It is a tool used to gather information regarding one or more objects from a distance. The drawbacks of this technique make it unfavorable to use, such as cost, permission in most cases, and the quality of the image "resolution" negotiable. Guimarães et al. [96] claimed that satellite data's spatial and temporal resolutions are often unsuitable for achieving regional or local forest objectives with traditional aerial and space-based SAR platforms. By contrast, even if their products have a more suitable spatial scale, manned aircraft are expensive when they are frequent. Time series monitoring is desirable. In addition, Xiang et al. [97] described the data from manned aircraft and satellite platforms as susceptible to cloudy sky conditions, which attenuate electromagnetic waves and lead to loss of information and data degradation.

4.3.3 UAVs

Ecke et al. [98] defined it as a drone or unmanned aircraft. A drone is a type of aircraft that operates with a remote control, or that is auto-programmed, and it has been used in various forest applications. It is important to note that the Department of National Defense and Canadian Armed Forces (DND/CAF) has exchanged the term "unmanned air vehicle" for "uncrewed air vehicle" to ensure gender-neutral terminology [11]. Accordingly, Seifert et al. [99] identified that drones use two types of sensors in a forest: the first is laser scanning or airborne LiDAR (ALS). Unmanned airborne vehicles are low-altitude remote sensing platforms, less affected by atmospheric factors during data acquisition. They offer the advantages of affordability, simple operations, fast imaging speed, and high spatial and temporal resolutions [49]. The second is image-based sensors UAV, which provide unprecedented spatial and temporal resolution imagery [100]. In Figure 7, the authors identified the satellite and UAV remote sensing techniques and represented several models and aspects connecting remote sensing with forest health. In Table 6, the authors compare the three remote sensing approaches from different perspectives: cost, permission, distance, weather conditions, temporal resolution, and spatial resolution. This table presents fundamental trends, but their manifestation may vary depending on the project's objectives, tools, and methodologies.



Figure 7: Sankey diagram for remote sensing.

	1	1	8
	Satellite	Manned Aircraft	Near-Surface Remote Sensing (UAV)
Cost	High	High	Low to Moderate
Permission	May required	Need permission	No need in most cases
Distance	Large area	Medium area for fuel reasons	Small area for batteries reasons
Weather Conditions	Low sensitive	Sensitive	Sensitive
Temporal Resolution	Moderate to High	Moderate to High	High
Spatial Resolution	Moderate to High	High	High

Table 6: Compares different techniques of remote sensing

UAV Applications UAV applications have become an increasingly popular forest monitoring and management tool, with applications ranging from wildfire prevention to tree damage detection. UAVs are valuable tools in studying forests and ecosystems—UAVs performed in several applications, such as UAVs for wildfires or UAVs for trees. First, the use of UAVs for wildfires has become a growing concern due to their severe impact on ecosystem degradation. Researchers have developed various techniques to address this issue, including using satellites, crewed aircraft, and UAVs to monitor and detect wildfires. Recently, UAVs have been a technique that researchers focus on to detect wildfires with the ability to cover a wider area and within an acceptable cost. A multi-model UAV-collected dataset of dual-feed side-by-side videos, including RGB and thermal images, has achieved higher accuracy than single-channel video feeds using a deep learning-based methodology [18].

Moreover, researchers emphasize the importance of studying forest fires and their impact on ecological degradation. Researchers suggest that the current observation of forest fires requires constant monitoring of all potential locations, particularly those with high fire risk [44]. UAVs have proven invaluable in monitoring and enhancing the quality of wildfire detection. In this regard, researchers proposed a color code identification, smoke motion recognition, and fire classification algorithm to improve the accuracy of detecting forest fires. Given the possibility of false alarms, the authors investigated methods to increase the accuracy of detecting forest fires [44].

Second, UAVs for trees from an ecological perspective can monitor weed vegetation, measure tree height, and estimate deforestation rates. Nowadays, many scientists use UAVs to investigate the status of the forest through tree characteristics, to evaluate soil moisture, or to enhance the quality of the citizen's property or public forest by developing irrigation techniques. Above-ground biomass (AGB) is a critical parameter for many environmental studies in reducing forest degradation. UAVs can measure the AGB in tropical mountain forests using Structure from Motion (SfM) with RGB sensors to estimate the tree height and the breast height (DBH) diameter, which are the inputs in calculating the AGB [46]. In addition, Brede et al. [54] studied the capability of estimating AGB using UAV-laser scanning. They

found that measuring the AGB for a single tree using traditional forest inventory methods that use allometric equations has low accuracy, is hard to implement, and is expensive. As a result, the author is investigating the ability to estimate AGB using UAV-LS.

The attributes of trees, including their height, canopy cover, and degree of defoliation, are important indicators of forest health, as they provide information on the ecological and environmental conditions of the ecosystem. Measuring the tree height for intensive forest monitoring using UAV-photogrammetric showed the ability to measure the tree height with the same accuracy level as field measurement [101]. In addition, UAVs can measure the physical attributes of Pinus halepensis trees [47]. Zhang et al. [49] tested the ability of UAV-Hyperspectral to measure the defoliation of trees during the Dendrolimus tabulaeformis Tsai et Liudisaster.

Other areas that UAVs have discovered in the forest ecosystem, such as Näsi et al. [5], identify bark beetle damage at the individual tree level of an urban forest. In addition, Camarretta et al. [13] studied the capability of active and passive sensors to measure the structure of the forest and the ability to restore it. Furthermore, Lu et al. [53] determined the effects of forests on revitalizing rural areas in five forest farms in Beijing, and Shin et al. [45] studied the capability of UAVs to measure the canopy fuels and forest structure in the United States. Lastly, the UAVs showed the ability to measure the physical attributes of a tree.

UAV Approaches UAVs are becoming more commonly used in forestry because of their advantages, such as spatial resolution, cost-effectiveness, adaptability, and more frequent visitations to relatively small areas. Ecke et al. [11] identified different approaches to UAVs: first, an image-based approach (passive approaches), which includes RGB, multispectral, hyperspectral, and thermal. Second is Laser scanning, or airborne LiDAR (active sensors). Figure 8 shows four different sensor types for the image-based approach.



Figure 8: UAV sensors type.

• Image-based approach (Passive approaches)

- RGB: a camera that can be connected to a drone flying to a high altitude and can capture various aspects by considering several steps, such as applying a filter to get a better result. In addition, the visible portion of the electromagnetic spectrum spans frequencies between 400 and 700 nm. The RGB, considering the preprocessing steps, will help researchers calculate plant vegetation.
- Thermal: Camera sensors for thermal imaging can detect infrared light with a wavelength between 7,500 and 13,500 nm. A thermal sensor can translate the observed energy into a temperature measurement. Modern cameras can detect numerous infrared energy bands, a capacity known as multispectral or hyperspectral. However, it frequently comes at the expense of reduced spatial resolution.
- Multispectral: In the 400–1000 nm range, multispectral sensors frequently measure certain "bands" of light, such as blue, green, red, red edge, and near-infrared. These bands will measure various vegetative traits, including stress and health. As a result, multispectral can assess the condition of a forest.
- Hyperspectral: Sensors can evaluate the vegetation level better than multispectral sensors, which can measure large amounts of data. The number of data that the hyperspectral can process increases the accuracy rate.

Table 7 summarizes the differences between the wavelengths of each image-based approach type. Wavelengths refer to the electromagnetic radiation (light) bands used for data collection and analysis. Wavelengths are important in drone-based remote sensing because they enable researchers to collect information regarding the Earth's surface and features.

• Laser scanning, or airborne LiDAR (active sensors)

Airborne laser scanning (ALS) can be used in forest inventory. Many countries around the world use ALS in forest inventories. Nevertheless, acquiring ALS data requires a degree of planning and investment, making these data sources cost-effective only on a large scale.

	Table 7: Sensor types and corresponding wavelengths						
Sensor Type	Wavelengths						
RGB	400–700 nanometers (nm) band						
Multispectral	400–1000 (nm) band						
Hyperspectral	Similar spectra as multispectral sensors but significantly differ in band numbers and widths.						
Thermal	7500 and 13,500 (nm) band						

UAV Parameters UAVs have significantly helped improve the techniques for measuring and monitoring forests. Accordingly, we illustrate several factors that affect the use of UAVs in monitoring or evaluating a forest. The UAV factors are summarized in these factors: altitude, overlap, speed, resolution, and weather conditions. Therefore, the success of UAVs in forest monitoring relies heavily on their flight parameters, including altitude, overlap, speed, resolution, and weather conditions. These factors play a crucial role in the accurate reconstruction and extraction of data. Recent studies have searched for the impact of these parameters on UAV image extractions, such as the research conducted by Seifert et al.[99]. Similarly, Tmušić et al. [102] identified a range of parameters that affect UAV flight time, such as weather conditions, payload, battery power or engine fuel, and UAV type. These findings highlight the importance of carefully considering UAV flight parameters when developing a forest health monitoring system. Table 8 summarizes several UAV parameters.

Table 8	: UAV	characteristics
10010 0		onunactoriotic

Paper	Overlap	Altitude	Flight Speed	Sensor Type	UAV Type
[47]	80%	100 m		RGB-Multispectra-Thermal	Mikrokopter OktoXL
[46]	90%	300 m	9 m/s	RGB-Multispectral	DJI Inspire I
[45]	85-90%	120 m	40–90 km/h	Multispectral	SenseFly eBee fixed-wing UAV
[48]	85%	90 m		Multispectral	A co-axial quad-copter UAV
[50]	75%	75 m	4 m/s	Multispectral	JI Matrice 600
[49]	70-60%	100 m		Hyperspectral	DJI Spreading Wings S1000+multi-rotor octocopter
[5]	75%	500 m		Hyperspectral	
[103]	90%	55 m	8 m/s	LiDAR-Hyperspectral	DJI M600 ProUAV
[104]	100%	50 m	2.0-4.0 m/s	Laser sensor	Riegl RiCOPTER with VUX-1UAV
[105]		40 /150 m	3.6 m/s	Laser sensor	Eight-rotor UAV
[54]		90 m		Laser sensor	VUX-1UA
[106]	90%	109 m		UAV with in-built true colour camera	JI Phantom 3 Professional quadcopter
[101]	80%	75 m			Octo XL 6 12 Octocopter mounted with a fixed lens
[55]	90-60%	300 m			DJI spreading wings S900

UAV Data Data acquisition approaches in UAVs consider different parts, such as the specific task to which the drone flies to achieve, determining the area of this task, weather conditions, and other parameters such as altitude (flight height), speed, and overlap (which help in improving image quality) [69]. The DJI UAV used Inspire 1 to collect data for 24 hours. The software was designed for precision flight to enhance flight characteristics. Data processing occurs after gathering the data and translating them into valuable data. Most of the review studies in this paper relied on commercial SfM software to create data products. Data analysis is a specific technique to measure and illustrate data to achieve a particular result, and different algorithms are considered in UAVs, such as the automatic tree detection and crown segmentation algorithm. Furthermore, machine learning has an impact on image recognition and tree classification. Figure 9 illustrates the process of monitoring a forest.



Figure 9: Overview of remote sensing system structure.

In summary, remote sensing has demonstrated the ability to monitor forests under different circumstances. However, relying on remote sensing to monitor forest health remains challenging. Dainelli et al. [107] clarified that in their reviewed papers, 60% of the studies in the entire dataset collected ground data. This type of data is gathered by carrying out field campaigns and measuring properties such as the size of trees, the species of trees present, and their health status. It is often also collected using GPS to help with the accuracy of imagery products. Collecting ground data takes

up many resources, such as time and money, and can be challenging over large areas and long periods. Further, Ecke et al. [11] define in their paper that drone surveys can help to simplify and sometimes even substitute for specific tasks in the field. However, drones' general replacement of ground surveys for FHM still needs more investigation to replace the field measurement. In Table 9, the authors illustrate several published papers regarding RS approaches by considering each paper's data acquisition techniques, tools, and aspects.

Paper	Application	Approach	Aspect	Sensor Type	Data Acquisition	Time-Frame
[40]	Tree	Image-based	Agriculture	Satellite	NASA	
[32]	Tree	Image-based	Agriculture	Satellite	NASA	5 Years
[56]	Tree	Image-based	Agriculture	Satellite	Synthetic Aperture Radar	1 Day
[57]	Tree		Tree Damage	Satellite	Global Navigation Satellite	4 months
[53]	Tree	Image-based	Forest Structure	Near-Surface	Multispectral	2004, 2009, 2014
[106]	Tree	Image-based	Vegetation	Drone	UAV data	1 Month
[9]	Tree	Image-based	Tree	Drone	UAV data	3 Days
[46]	Tree	Image-based	Tree	Drone	UAV data	
[55]	Tree	Image-based	Leaf Index	Drone	UAV data	20 Days
[49]	Tree	Image-based	Tree Damage	Drone	UAV data	1 Month
[44]	Wildfire	Image-based	FireRisk	Drone	UAV data	

 Table 9: Summarizes remote sensing measurement approaches

UAV Features, Drawbacks, and Challenges Advances in UAV technology are rapidly being adopted and implemented for forests. As with any technology, UAVs offer benefits and limitations for forest monitoring and management. Flexibility, relatively low costs, and the possibility of flying below the cloud cover. Furthermore, it provides a unique spatial resolution and angle of view data and can offer lower ground sample distances (GSDs) [11]. In addition, compared to traditional inventory surveys, UAVs cause less disturbance to the sample area's flora and animals [106].

UAV observations also provided a unique insight and a historical record with the capacity for near real-time reporting and validation of change events and processes in a forest. Despite the many advantages of UAVs, some shortcomings are found. Ecke et al. [11] identified these drawbacks; the long-term monitoring of forests needs to be better presented. The data pipeline from acquisition to final analysis often relies on commercial software at the expense of open-source tools. In addition, other drawbacks related to the hardware include flights such as battery duration, payload weight, and sensitivity to terrible weather conditions. Further, several challenges have been addressed in the UAV approach, such as the lack of accuracy in detecting tree height, measuring AGB, and determining the flight time of the UAV. Other challenges include the limitation of accessing underground attributes and battery duration. Despite all of the critical challenges listed above, the advantages of using UAVs instead of other remote sensing platforms far outweigh the drawbacks if used appropriately.

5 MULTI DATA SOURCES

The multi-data sources approach in forest health monitoring uses different types of sensors to gather data about certain objectives. This technique gives researchers and forest managers better knowledge of a forest as it can cover various aspects. Some common types of sensors are used in forest health monitoring (remote sensing sensors and ground-based sensors). Remote sensing sensors, such as satellites, aircraft, or drones, can capture images and data regarding forests from above. Conversely, Ground-based sensors are installed in a forest to capture environmental data. Combining different types of these techniques or combining the same tools with varying types of sensors can lead to a better version of forest monitoring.

Multi-data sources depend on integrating different tools, sensor types, or data collection methods. There are several general techniques commonly used in the multi-data sources approach. One of these techniques is in-site, which collects data regarding different local environments. Another approach is remote sensing, which detects an object from above. The third approach is data fusion, which combines data from multiple sensors and sources to monitor a specific aspect of a forest. Ultimately, machine learning can analyze, classify, and predict particular events from multi-data source systems.

Data fusion is the third approach, and this technique has provided a wide range of vision in forest health monitoring in recent years. Data fusion combines data from multiple sources to achieve a goal in a specific aspect. Data fusion has been defined by Hall & Llinas [108] as "data fusion techniques that combine data from multiple sensors, and related information from associated databases, to achieve improved accuracies and more specific inferences than could be achieved by using a single sensor alone." In addition, they defined the levels of fusing the data. The first level is a row of sensor data when the sensor data are commensurate. However, when they are not commensurate, we move to the other two levels, which work with the same feature/state vector level or decision level. To illustrate the idea, we provide an example of row sensor data when gathering information for the same object, such as two visual images. Another

example of a feature level is when you extract a feature from this object, such as tree crown detection. An example of decision-level fusion methods example is weighted decision methods (voting techniques).

Data fusion and multi-sensor techniques showed the ability to monitor or create a decision support system. For example, Yang et al. [109] generated a multi-system data fusion model that can analyze drought-induced mortality in a temperate forest by estimating daily 30-m resolution evapotranspiration (ET) and the associated Evaporative Stress Index (ESI). The paper studied the techniques in two different scales and used different satellite and measurement techniques to detect and predict forest mortality due to drought. Further, multi-sensors can investigate the relationships between flood severity, precipitation, and deforestation, which can increase the ability to understand the reason behind floods and can reduce the damages [59]. The authors applied regression analysis techniques to measure the impact of precipitation and deforestation on floods. The authors found that there is a significant relationship between precipitation and flood. At the same time, flood and deforestation have no tangible relation.

Besides mortality and flood, tree diversity and species are other signs of forest health. UAVs and satellites have played a role in enhancing the capability of detecting tree species. For example, Hartling et al. [110] used data fusion and machine learning methods to investigate UAV-based multi-sensory ability to classify tree species in an urban environment. The authors have explained the techniques for gathering data from UAVs with different types of sensors, such as hyperspectral and multispectral. They also explained the role of machine learning random forest and Support vector machine in this process. In addition, the authors provided the overall accuracy of this process in specifying the classification of the trees and identifying the limitations and future work in this area. Moreover, Host et al. [111] investigated the presence and abundance of ash trees in Minnesota forests. The authors used different satellite techniques and combined a Landsat image with LiDAR to determine the existence of an ash tree. As a result, the authors found that the overall accuracy of detecting an ash tree is 64% for all ash species and 72% for black ash, and accuracy increased with the length of the time series. Table 10 identifies several methods and aspects for using data fusion in forest health monitoring.

Table 10: Summarizes published papers regarding data fusion

Paper	Sensor Type	Aspect	Location
[110]	UAV- Different approaches	Tree Species	USA
[112]	Field - Satellite - UAS (Drone)	Forest Disturbances (Diseases)	USA
[113]	Field - Satellite - UAV (Drone)	Crop Management	USA
[114]	Different Satellite	Tree Species	USA
[109]	Different Satellite	Tree Mortality	USA
[59]	Different Satellite	Flood Severity	Cambodia
[115]	Different Satellite	Crop Management	Vietnam - Lebanon

Crop management is another field in which data fusion has increased the potential to work. Exploring the ability of UAVs and satellites to measure soybean LAI, AGB, and leaf nitrogen concentration (N) [113]. The authors gathered information regarding the canopy spectral information with canopy structure features using satellites and UAVs. The authors applied different techniques (partial least squares regression, random forest regression, support vector regression, and extreme learning regression with a newly proposed activation function) separately to UAV and satellite and then integrated them to measure the accuracy of monitoring crops. As a result, the authors found that the overall accuracy increased when they combined the UAV with satellite techniques rather than using them alone.

6 DISCUSSION

In this section, we synthesize the findings from this paper on the application of forest health. We identify common themes and key insights for each section.

- Forest Health: The research has yielded valuable insights into the complexity of developing a forest index or monitoring system, underscoring the significance of considering various factors. These factors encompass the type of forest, geographical location, elevation, distance from roads, and local climatic conditions. Equally critical is the objective of the index, project timelines, and the tools employed for its creation. When establishing a forest index, thoughtful consideration of these elements positively influences ecosystem health, reduces the risk of forest fires, and enhances habitat quality for diverse organisms. This, in turn, facilitates informed future planning based on historical data. For instance, Winarso et al.[36] suggested that a satellite-based mangrove index is an alternative to the widely utilized NDVI. It is important to note a finding: no comprehensive framework or guideline exists for generating an index that characterizes a forest's condition. However, as mentioned before, some researchers investigate techniques to generate an index.
- Measurement Approaches: The authors have presented a review of the usage of each measurement and provide several examples of using these technologies in studying forests. Figure 10 shows that remote sensing

has garnered the highest number of reference articles, highlighting the significance of this technology in modern research. Extensive references demonstrate the widespread use and recognition of remote sensing as a powerful tool for studying various aspects of the environment. Remote sensing has shown its remarkable ability to investigate diverse areas, including but not limited to tree damage assessment and wildfire monitoring. These findings emphasize remote sensing's invaluable role in enabling researchers to gather crucial data from vast and inaccessible regions, offering unparalleled insights into ecological changes and natural disasters.



Figure 10: Techniques and aspects.

First, field observation is one of the techniques that researchers and scientists rely on during forest investigation. We found that field observation has been used in most of the reviewed papers for collecting data or validation. Field observations have proven their role in generating accurate forest health by showing the ability to measure various indicators, including biological or ecological. For example, Ranau et al. [60] used the bird as a biological indicator to understand the quality of specific areas. However, finding a recent research paper describing field observation takes time and effort.

Second, ground-based sensors are another technique used in studying forest health. This technique facilitates informed decision-making by enabling stakeholders to implement protective measures and enhance their comprehension of forest ecosystems. We used various published studies covering the importance of ground-based sensors in studying forest health to highlight the knowledge of the impact of this approach in studying forests. For example, Marsh et al. [23] showed that the relationship between canopy cover, forest structure, and microclimate leads to understanding the degradation of forests and expecting climate change. This technique has indisputable value in expanding our comprehension of forest health evaluation. Third, the remote sensing technique stands among the forefront methodologies driving research in forest health assessment. This technique provides the ability to investigate several factors and attributes of forests. Sudhakar et al. [44] studied the ability of UAVs to enhance the warning system for fire risk. In summary, remote sensing has demonstrated undeniable utility in advancing our understanding of forest health assessment, and researchers suggest collaborating remote sensing with field observations can enhance the technique of studying forest health from several perspectives.

Furthermore, as the exact figure, ground-based sensors and field observations are vital components in forest health monitoring. These methods offer distinct advantages, allowing researchers to measure forest health accurately. Ground-based approaches provide valuable on-the-ground data that complement and validate remote sensing findings, contributing to a more comprehensive and reliable assessment of forest ecosystems.

Figure 11 complements the findings by visually representing the geographical distribution of the article references. Each country's proportionate presence on the graph reflects scientific interest and involvement in studying various environmental aspects. The varying sizes of the bars corresponding to different countries signify the relative emphasis and contribution of each nation's research efforts in advancing the field of environmental monitoring and management. As we observe in that figure, the distribution percentages show that this review's findings have wide-reaching implications for global conservation strategies and sustainable practices in diverse forest ecosystems worldwide.

Multi Data Sources: Utilizing multiple sensors provides researchers and forest managers with a comprehensive
understanding of forest dynamics, enabling the assessment of various critical aspects. For instance, Yang
et al. [109] introduced a data fusion model capable of analyzing drought-induced mortality in temperate
forests. This model estimates daily 30 m resolution evapotranspiration (ET) and the associated ESI. In addition,
multi-sensor technology proves valuable in studying natural phenomena such as floods, as demonstrated by



Figure 11: Geographical locations.

Kim et al. [59]. Collectively, these studies underscore the immense potential of multi-data source systems in developing forest indexes and their applications.

7 Research Challenges and Directions

The increasing trend of forest health demands a prediction, evaluation, and monitoring strategy. Considering the various factors that affect forest health, modeling forest health with more than one contributing factor will help develop an assessment, prediction, or warning system. Several approaches and techniques exist for creating an index that can observe and quantify various indicators, ranging from avifauna diversity to overall ecosystem vitality. In addition, deciding which tools, techniques, and indicators to use is challenging. Thus, in this section, the authors will clarify some discussed gaps in existing research and will identify challenges. In Section 7.1, the authors will describe the difficulties in determining the indicators. In Section 7.2, we will cover the challenges of the measurement approaches. Lastly, in Section 7.3, we will highlight the impact of the time frame in an outdoor environment.

7.1 Standardization

From our knowledge perspective, we found difficulties in exploring the indicators that are used to study forest health. Because of the presence of many factors that affect the health of forests, the authors found it a great challenge to know these indicators and determine the appropriate indicators to study forest health in a particular area. Accordingly, there are no well-accepted methodologies used to measure forest health. Some frameworks, such as (foresthealthindex.org) or the USDA reporter, highlight it [4]. Many of these factors encompass ecological, meteorological, and biological indicators, offering unique insights into forest health. Combining two or more indicators can yield an index that provides a view of forest quality. For example, Marsh et al. [23] generated a model that studies the correlation between forest structure and air temperature. However, the challenges do not end with the identification of the indicators. Selecting the most suitable experimental plots is an equally formidable challenge, given the many of factors that must be carefully considered beforehand, including securing the necessary permissions.

Therefore, several approaches can be considered to get over these challenges. Expert insights can help in navigating the intricate landscape of forest health assessment. Additionally, researchers can draw inspiration from various websites and organizations dedicated to forest health research, leveraging the knowledge and methodologies already developed and tested in the field. By organizing our discussion in this manner, we aim to provide a structured exploration of the challenges faced in forest health assessment, starting with identifying indicators and concluding with the practical considerations of selecting suitable experimental plots.

7.2 Measurement Approaches

Another challenge in studying forest health assessment is determining the tools or approaches. Several approaches exist to check forest health, including field observations, ground-based sensors, remote sensing, and multi-sensor methods integrating more than one sensor. Each tool has features, drawbacks, and challenges. This part will discuss these approaches.

Despite the importance of field observation, several challenges can be detected in this technique, including accessibility, time, human mistakes, and cost constraints. Accessibility, time-consuming, and a high chance of a human mistake;

we all know that there is a chance that human eyes or ears could record something that is not right—finally, the cost constraints for preparing and sending a research group to a forest. As a result, Park et al. [9] identified that measuring tree crowns for individual tree leaves is doable but labor-intensive. In addition, it is hard to accurately assess the tree attributes in tall, dense, multi-layered forests. Further, Ampatzidis & Partel, [50] claimed that evaluating the field's phenotype from these perspectives is labor-intensive and time-consuming, mainly when covering large areas. In contrast, the ability to use remote sensing approaches such as a satellite or drone could help to overcome some of these challenges as it showed the ability to measure tree crowns with different accuracy rates.

Since ground-based sensor is a crucial technique, it is essential to acknowledge and address their challenges, such as sensor placement, maintenance, power supply, and cost constraints. All of these challenges are impacting the decision to use this technique. The first challenge involves the difficulty of building a station or installing and maintaining the sensors in an outdoor environment. Other risks include the attacks of bugs and animals and harsh weather conditions. It could also require special permission to install the sensors. The second challenge is to consider the impact of ground-based sensors on the growth of trees or plants. The third challenge is the cost of creating and maintaining a station to monitor the forest's health. Finally, network coverage plays a crucial role in the ground-based approach. For example, regarding budgetary restrictions, Marsh et al. [23] reduced the number of air temperature data loggers during the experiment. However, these challenges can be solved one way or another depending on the importance and value this approach adds to the project. For example, generating a wireless sensor network can solve the problem [116]. As a future direction, several potential works could be done in this field, such as investigating the ability to generate a wireless sensor network in rural areas. Developing a ground-based sensor to monitor illegal logging using the sound of chainsaws.

Over the past decade, remote sensing has revolutionized the ability to monitor forests with an acceptable accuracy rate on a large spatial scale. But it also comes with several challenges, such as cost, accessibility, and other technical challenges regarding spatial and temporal resolution or the battery duration and payload weight with a drone. Ecke et al. [11] identified several challenges and future work regarding remote sensing. One of these future works is looking at the flight parameters, which requires more attention from the author's perspective during influence data quality. Highlighting that the resolution ability of the satellite remote sensing for monitoring forest ecosystems sometimes needs to be improved during the limits of feasibility [10]. Moreover, Lausch et al. [1] explored that linking terrestrial and remote sensing-based approaches is an issue that needs to be addressed, which could provide a better vision of monitoring a forest.

Multi-data source is a technique researchers have provided as a future direction of forest health monitoring, with the capability to cover several issues, such as enhancing the accuracy or reducing the cost. Still, various challenges are addressed, such as data integration, temporal synchronization, data fusion, and model selection. Data integration requires understanding the data format from different sensors and dealing with temporal synchronization, which refers to gathering the data at the exact time. Data fusion and model selection refer to the difficulty of choosing the methods and techniques for a multi-sensor system. So, regarding the aforementioned, Hartling et al. [110] addressed the issues of applying multiple UAV sensors for tree species to create a robust training model that can be used in multiple locations. Also, there are several challenges in accurately engaging data fusion methods between UAV and other sensors; for example, collecting imagery from both data sources on the same date [112].

7.3 Data Collection

In the context of outdoor environments, especially within forest ecosystems, the timing of a research project emerges as a crucial factor warranting closer scrutiny. The temporal dimensions of data collection substantially influence various facets of the research process, encompassing project budgeting and experimental planning. Take, for instance, the investigation by Han et al. [26], which delved into breeding areas within forest ecosystems. Their experiment, conducted in late March, illustrates the critical role of timing, particularly when studying breeding activities. The timing directly affects the availability and behavior of the species under investigation. If a research team must synchronize their data collection with the breeding season, any deviation from the optimal timing might necessitate postponing experiments for nearly a year. Such delays can significantly impact project timelines and resource allocation.

Another illustration of data collection constraints arises when researchers study the effects of seasonal variations in vegetation levels. Such investigations requiring a long period of time necessitate meticulous planning to capture the subtle yet critical changes that unfold over time. This type of project needs to be considered, especially for researchers who have a specific period of time to finish the project. In addition, adhering to the appropriate timing can have profound implications for project timelines and resource allocation.

Researchers can explore various strategies to mitigate these temporal challenges, including leveraging existing datasets. As demonstrated by Han et al. [26] in their previous work, tapping into data from previously published studies to

support their collected data makes it more valuable. This approach offers a pragmatic solution and contributes to the accumulation of knowledge within the research community. Lastly, researchers need to expect bias and damage that can happen during data collection.

8 CONCLUSIONS

Forests, as the habitats for most land mammals, birds, insects, and plants, play a crucial role in supporting life on our planet. In addition, the well-being of a substantial portion of the global population is intricately tied to forests and their resources [117]. In light of the increasing challenges posed by wildfires and deforestation, scientists and decision-makers have intensified their efforts toward developing effective forest health monitoring systems. However, creating such monitoring systems in outdoor environments has complexities, demanding both time and financial resources. Consequently, there is a growing need to explore innovative methodologies that enhance the accuracy of forest health monitoring while minimizing time and cost implications.

This review examined the FHI concept, highlighting the various factors influencing its development. Furthermore, it delved into multiple indicators that offer insights into forest structure and condition. In addition to this, it explored diverse techniques and approaches for establishing forest health monitoring systems. Among these, field observations are a valuable tool, although they are hindered by limitations such as restricted access to certain areas at specific times. Another approach involves the deployment of ground-based sensors within the forest environment. Lastly, remote sensing methods, including satellite imagery, crewed aircraft, and drones, have demonstrated significant potential in assessing forest health, particularly concerning ecological indicators such as tree damage, height, and vegetation cover. However, it is still hard to rely on remote sensing techniques to determine the health status of an area.

The findings of this review have identified several areas for improvement in current research. One notable gap concerns the temporal scope of experiments, with many studies focusing on a single year's data [118]. Researchers are also actively exploring incorporating classification filters and textural features to automate the identification process, aiming for improvements [49]. In addition, enhancing the accuracy of canopy height models presents a promising avenue for future research [105]. To address limitations in dense canopy areas, researchers are considering strategies such as increasing point cloud density through repeated flights and optimizing flight paths to maximize trunk visibility [54]. Lastly, harmonizing multimode data fusion with practical operational feasibility remains a critical research challenge [112]. In light of these research gaps, an intriguing avenue for future exploration involves the feasibility of integrating various techniques to construct a robust FHI utilizing multiple sensors. This approach can overcome some of the challenges outlined in this review. I want to emphasize that open questions persist in forest health monitoring and forest health indexing. Furthermore, it is worth noting that although many studies describe the capabilities of monitoring tools for specific aspects of forest health, only a few integrate these findings directly into the broader context of forest monitoring or forest index. In conclusion, the research reviewed here underscores the critical importance of preserving the health of our forests, given their profound ecological and societal significance. The road ahead involves pursuing innovative approaches that leverage the strengths of various monitoring techniques to create a comprehensive and cost-effective forest health assessment system. As we navigate these challenges, it is clear that exploring these open questions will continue to shape the future of forest health index research.

References

- Angela Lausch, Stefan Erasmi, Douglas King, Paul Magdon, and Marco Heurich. Understanding forest health with remote sensing -part i—a review of spectral traits, processes and remote-sensing characteristics. *Remote Sensing*, 8:1029, 12 2016.
- [2] Susan Trumbore, Paulo Brando, and Henrik Hartmann. Forest health and global change. *Science*, 349(6250):814–818, 2015.
- [3] Dayle D. Bennett and Borys M. Tkacz. Forest health monitoring in the united states: a program overview. *Australian Forestry*, 71:223–228, 1 2008.
- [4] KaDonna C Randolph. Forest ecosystem health indicators. https://usfs.maps.arcgis.com/apps/ MapJournal/index.html?appid=a434a8d7b3d447c8a747efb4fd22d742#, 2008. [Online; accessed 26-July-2023].
- [5] Roope Näsi, Eija Honkavaara, Minna Blomqvist, Päivi Lyytikäinen-Saarenmaa, Teemu Hakala, Niko Viljanen, Tuula Kantola, and Markus Holopainen. Remote sensing of bark beetle damage in urban forests at individual tree level using a novel hyperspectral camera from uav and aircraft. Urban Forestry & Urban Greening, 30:72–83, 2018.

- [6] Langning Huo, Eva Lindberg, and Henrik Persson. Normalized projected red and swir (nprs): A new vegetation index for forest health estimation and its application on spruce bark beetle attack detection. pages 4618–4621. Institute of Electrical and Electronics Engineers Inc., 9 2020.
- [7] Angela Lausch, Erik Borg, Jan Bumberger, Peter Dietrich, Marco Heurich, Andreas Huth, András Jung, Reinhard Klenke, Sonja Knapp, Hannes Mollenhauer, Hendrik Paasche, Heiko Paulheim, Marion Pause, Christian Schweitzer, Christiane Schmulius, Josef Settele, Andrew Skidmore, Martin Wegmann, Steffen Zacharias, Toralf Kirsten, and Michael Schaepman. Understanding forest health with remote sensing, part iii: Requirements for a scalable multi-source forest health monitoring network based on data science approaches. *Remote Sensing*, 10, 7 2018.
- [8] Angela Lausch, Stefan Erasmi, Douglas J King, Paul Magdon, and Marco Heurich. Understanding forest health with remote sensing-part ii—a review of approaches and data models. *Remote Sensing*, 9(2):129, 2017.
- [9] John Y Park, Helene C Muller-Landau, Jeremy W Lichstein, Sami W Rifai, Jonathan P Dandois, and Stephanie A Bohlman. Quantifying leaf phenology of individual trees and species in a tropical forest using unmanned aerial vehicle (uav) images. *Remote Sensing*, 11(13):1534, 2019.
- [10] Robert Minařík, Jakub Langhammer, and Theodora Lendzioch. Automatic tree crown extraction from uas multispectral imagery for the detection of bark beetle disturbance in mixed forests. *Remote Sensing*, 12(24), 2020.
- [11] Simon Ecke, Jan Dempewolf, Julian Frey, Andreas Schwaller, Ewald Endres, Hans-Joachim Klemmt, Dirk Tiede, and Thomas Seifert. Uav-based forest health monitoring: a systematic review. *Remote Sensing*, 14, 7 2022.
- [12] Pablo Torres, Marina Rodes-Blanco, Alba Viana-Soto, Hector Nieto, and Mariano García. The role of remote sensing for the assessment and monitoring of forest health: A systematic evidence synthesis. *Forests*, 12, 8 2021.
- [13] Nicolò Camarretta, Peter A Harrison, Tanya Bailey, Brad Potts, Arko Lucieer, Neil Davidson, and Mark Hunt. Monitoring forest structure to guide adaptive management of forest restoration: a review of remote sensing approaches. *New Forests*, 51:573–596, 2020.
- [14] Marija Ćosović, Miguel Bugalho, Dominik Thom, and José Borges. Stand structural characteristics are the most practical biodiversity indicators for forest management planning in europe. *Forests*, 11:343, 3 2020.
- [15] Marion Pause, Christian Schweitzer, Michael Rosenthal, Vanessa Keuck, Jan Bumberger, Peter Dietrich, Marco Heurich, András Jung, and Angela Lausch. In situ/remote sensing integration to assess forest health—a review. *Remote Sensing*, 8:471, 6 2016.
- [16] Sarah JK Frey, Adam S Hadley, and Matthew G Betts. Microclimate predicts within-season distribution dynamics of montane forest birds. *Diversity and distributions*, 22(9):944–959, 2016.
- [17] Roshani, Haroon Sajjad, Tamal Kanti Saha, Md Hibjur Rahaman, Md Masroor, Yatendra Sharma, and Swades Pal. Analyzing trend and forecast of rainfall and temperature in valmiki tiger reserve, india, using non-parametric test and random forest machine learning algorithm. *Acta Geophysica*, 71:531–552, 12 2022.
- [18] Xiwen Chen, Bryce Hopkins, Hao Wang, Leo O'Neill, Fatemeh Afghah, Abolfazl Razi, Peter Fule, Janice Coen, Eric Rowell, and Adam Watts. Wildland fire detection and monitoring using a drone-collected rgb/ir image dataset. *IEEE Access*, 10:121301–121317, 2022.
- [19] E R Arwanda and R Safe'i. Assessment of forest health status of panca indah lestari community plantation forest (case study in bukit layang village, bakam district, bangka regency, bangka belitung province). *IOP Conference Series: Earth and Environmental Science*, 918:012031, 11 2021.
- [20] Rahmat Safe'i, Arief Darmawan, Hari Kaskoyo, and Citra Farshilia Gayansa Rezinda. Analysis of changes in forest health status values in conservation forest (case study: plant and animal collection blocks in wan abdul rachman forest park (tahura war)). In *Journal of Physics: Conference Series*, volume 1842, page 012049. IOP Publishing, 2021.
- [21] Jun Yang, Guanghui Dai, and Shurong Wang. China's national monitoring program on ecological functions of forests: An analysis of the protocol and initial results. *Forests*, 6:809–826, 3 2015.
- [22] Qinghai Song, Chenna Sun, Yun Deng, He Bai, Yiping Zhang, Hui Yu, Jing Zhang, Liqing Sha, Wenjun Zhou, and Yuntong Liu. Tree surface temperature in a primary tropical rain forest. *Atmosphere*, 11:798, 7 2020.
- [23] Christopher D Marsh, Ross A Hill, Matthew G Nowak, Emma Hankinson, Abdullah Abdullah, Phillipa Gillingham, and Amanda H Korstjens. Measuring and modelling microclimatic air temperature in a historically degraded tropical forest. *International Journal of Biometeorology*, 66(6):1283–1295, 2022.

- [24] Dmitry Pershin, Natalia Malygina, Dmitry Chernykh, Roman Biryukov, Dmitry Zolotov, and Lilia Lubenets. Variability in snowpack isotopic composition between open and forested areas in the west siberian forest steppe. *Forests*, 14(1):160, 2023.
- [25] Jack D Shutt, Sophie C Bell, Fraser Bell, Joan Castello, Myriam El Harouchi, and Malcolm D Burgess. Territorylevel temperature influences breeding phenology and reproductive output in three forest passerine birds. *Oikos*, 2022(8):e09171, 2022.
- [26] Xuesong Han, Yumin Guo, Chunrong Mi, Falk Huettmann, and Lijia Wen. Machine learning model analysis of breeding habitats for the black-necked crane in central asian uplands under anthropogenic pressures. *Scientific reports*, 7(1):6114, 2017.
- [27] Daichi Iijima, Atsushi Kobayashi, Gen Morimoto, Masami Hasegawa, Seiya Abe, and Masashi Murakami. A trait-based approach to seasonal dynamics of an alpine and subalpine passerine bird assemblage. *Journal of Ornithology*, 163(3):709–721, 2022.
- [28] Hao Wu, Jiehua Li, Jie Zhou, Mingxi Jiang, and Xinzeng Wei. Elevational pattern and temperature sensitivity of spring leaf phenology of three co-occurring tree species in a subtropical mountain forest. *Trees*, pages 1–12, 2023.
- [29] Marlene L Huerta, Noah P Molotch, and James McPhee. Snowfall interception in a deciduous nothofagus forest and implications for spatial snowpack distribution. *Hydrological Processes*, 33(13):1818–1834, 2019.
- [30] Chunrong Mi, Falk Huettmann, and Yumin Guo. Obtaining the best possible predictions of habitat selection for wintering great bustards in cangzhou, hebei province with rapid machine learning analysis. *Chinese Science Bulletin*, 59:4323–4331, 2014.
- [31] G Mancino, A Nolè, F Ripullone, and A Ferrara. Landsat tm imagery and ndvi differencing to detect vegetation change: assessing natural forest expansion in basilicata, southern italy. *iForest Biogeosciences and Forestry*, 7:75–84, 4 2014.
- [32] John D Bolten, Wade T Crow, Xiwu Zhan, Thomas J Jackson, and Curt A Reynolds. Evaluating the utility of remotely sensed soil moisture retrievals for operational agricultural drought monitoring. *IEEE Journal of Selected Topics in Applied Earth Observations and Remote Sensing*, 3(1):57–66, 2009.
- [33] Nathalie JJ Bréda. Ground-based measurements of leaf area index: a review of methods, instruments and current controversies. *Journal of experimental botany*, 54(392):2403–2417, 2003.
- [34] Graham Pope and Paul Treitz. Leaf area index (lai) estimation in boreal mixedwood forest of ontario, canada using light detection and ranging (lidar) and worldview-2 imagery. *Remote Sensing*, 5:5040–5063, 10 2013.
- [35] I Olthof and DJ King. Development of a forest health index using multispectral airborne digital camera imagery. *Canadian Journal of Remote Sensing*, 26(3):166–176, 2000.
- [36] Gathot Winarso, Muhammad Kamal, Syamsu Rosid, Wikanti Asriningrum, and Jatna Supriyatna. Correlation analysis of satellite-based mangrove index and mangrove forest health at segara anakan cilacap, central java, indonesia. *IOP Conference Series: Earth and Environmental Science*, 500:012050, 6 2020.
- [37] Jan Mišurec, Veronika Kopačková, Zuzana Lhotáková, Jan Hanuš, Jörg Weyermann, Petya Entcheva-Campbell, and Jana Albrechtová. Utilization of hyperspectral image optical indices to assess the norway spruce forest health status. *Journal of Applied Remote Sensing*, 6(1):063545–063545, 2012.
- [38] Shiekh Marifatul Haq, Muhammad Shoaib Amjad, Muhammad Waheed, Rainer W. Bussmann, and Jarosław Proćków. The floristic quality assessment index as ecological health indicator for forest vegetation: A case study from zabarwan mountain range, himalayas. *Ecological Indicators*, 145:109670, 12 2022.
- [39] Forest Health Index. https://foresthealthindex.org/. Accessed: 2023-06-21.
- [40] Inbal Becker-Reshef, Chris Justice, Mark Sullivan, Eric Vermote, Compton Tucker, Assaf Anyamba, Jen Small, Ed Pak, Ed Masuoka, Jeff Schmaltz, et al. Monitoring global croplands with coarse resolution earth observations: The global agriculture monitoring (glam) project. *Remote Sensing*, 2(6):1589–1609, 2010.
- [41] Shulin Chen, Zuomin Wen, Hong Jiang, Qingjian Zhao, Xiuying Zhang, and Yan Chen. Temperature vegetation dryness index estimation of soil moisture under different tree species. *Sustainability*, 7(9):11401–11417, 2015.
- [42] Sajjad Ahmad, Ajay Kalra, and Haroon Stephen. Estimating soil moisture using remote sensing data: A machine learning approach. *Advances in water resources*, 33(1):69–80, 2010.
- [43] Jakob Iglhaut, Carlos Cabo, Stefano Puliti, Livia Piermattei, James O'Connor, and Jacqueline Rosette. Structure from motion photogrammetry in forestry: A review. *Current Forestry Reports*, 5:155–168, 2019.

- [44] S Sudhakar, Varadarajan Vijayakumar, C Sathiya Kumar, V Priya, Logesh Ravi, and V Subramaniyaswamy. Unmanned aerial vehicle (uav) based forest fire detection and monitoring for reducing false alarms in forest-fires. *Computer Communications*, 149:1–16, 2020.
- [45] Patrick Shin, Temuulen Sankey, Margaret Moore, and Andrea Thode. Evaluating unmanned aerial vehicle images for estimating forest canopy fuels in a ponderosa pine stand. *Remote Sensing*, 10:1266, 8 2018.
- [46] Víctor González-Jaramillo, Andreas Fries, and Jörg Bendix. Agb estimation in a tropical mountain forest (tmf) by means of rgb and multispectral images using an unmanned aerial vehicle (uav). *Remote Sensing*, 11(12), 2019.
- [47] Filippo Santini, Shawn C Kefauver, Victor Resco de Dios, José L Araus, and Jordi Voltas. Using unmanned aerial vehicle-based multispectral, rgb and thermal imagery for phenotyping of forest genetic trials: A case study in pinus halepensis. *Annals of Applied Biology*, 174(2):262–276, 2019.
- [48] Jonathan P Dash, Grant D Pearse, and Michael S Watt. Uav multispectral imagery can complement satellite data for monitoring forest health. *Remote Sensing*, 10(8):1216, 2018.
- [49] Ning Zhang, Xiaoli Zhang, Guijun Yang, Chenghao Zhu, Langning Huo, and Haikuan Feng. Assessment of defoliation during the dendrolimus tabulaeformis tsai et liu disaster outbreak using uav-based hyperspectral images. *Remote Sensing of Environment*, 217:323–339, 2018.
- [50] Yiannis Ampatzidis and Victor Partel. Uav-based high throughput phenotyping in citrus utilizing multispectral imaging and artificial intelligence. *Remote Sensing*, 11(4):410, 2019.
- [51] Ronald E McRoberts, Warren B Cohen, Erik Naesset, Stephen V Stehman, and Erkki O Tomppo. Using remotely sensed data to construct and assess forest attribute maps and related spatial products. *Scandinavian Journal of Forest Research*, 25(4):340–367, 2010.
- [52] Lei Wang, Huaguo Huang, and Youqing Luo. Remote sensing of insect pests in larch forest based on physical model. In 2010 IEEE International Geoscience and Remote Sensing Symposium, pages 3299–3302. IEEE, 2010.
- [53] Shasha Lu, Yi Zhou, Haisheng Sun, Ni Chen, and Xingliang Guan. Examining the influencing factors of forest health, its implications on rural revitalization: A case study of five forest farms in beijing. *Land Use Policy*, 102, 2021.
- [54] Benjamin Brede, Kim Calders, Alvaro Lau, Pasi Raumonen, Harm M Bartholomeus, Martin Herold, and Lammert Kooistra. Non-destructive tree volume estimation through quantitative structure modelling: Comparing uav laser scanning with terrestrial lidar. *Remote Sensing of Environment*, 233:111355, 2019.
- [55] Dafeng Zhang, Jianli Liu, Wenjian Ni, Guoqing Sun, Zhiyu Zhang, Qinhuo Liu, and Qiang Wang. Estimation of forest leaf area index using height and canopy cover information extracted from unmanned aerial vehicle stereo imagery. *IEEE Journal of Selected Topics in Applied Earth Observations and Remote Sensing*, 12(2):471–481, 2019.
- [56] ES Mohamed, Abdelraouf Ali, Mohammed El-Shirbeny, Khaled Abutaleb, and Sameh M Shaddad. Mapping soil moisture and their correlation with crop pattern using remotely sensed data in arid region. *The Egyptian Journal of Remote Sensing and Space Science*, 23(3):347–353, 2020.
- [57] P S Anwar, R Safe'i, and A Darmawan. Landscape characteristics on forest health measurement plots in several forest functions. *IOP Conference Series: Earth and Environmental Science*, 950:012101, 1 2022.
- [58] Yasar Guneri Sahin. Animals as mobile biological sensors for forest fire detection. *Sensors*, 7(12):3084–3099, 2007.
- [59] Sangpil Kim, Hong-Gyoo Sohn, Mi-Kyeong Kim, and Hyongki Lee. Analysis of the relationship among flood severity, precipitation, and deforestation in the tonle sap lake area, cambodia using multi-sensor approach. *KSCE Journal of Civil Engineering*, 23:1330–1340, 2019.
- [60] George Hubert Petol, Loraiti Lolin, Vivian Rudolf, Fernandez Joel, and Marshall Neo Petol. Avifaunal survey of bukit balingkadus, a small fragmented forest in ranau, sabah, malaysia. *Journal of Tropical Biology & Conservation (JTBC)*, 19:109–123, 2022.
- [61] Cici Doria, Rahmat Safe'i, Dian Iswandaru, and Hari Kaskoyo. Fauna biodiversity as one of repong damar forest health indicators. In *IOP Conference Series: Earth and Environmental Science*, volume 886, page 012036. IOP Publishing, 2021.
- [62] Pham Quang Thu, Dao Ngoc Quang, Nguyen Minh Chi, Tran Xuan Hung, Le Van Binh, and Bernard Dell. New and emerging insect pest and disease threats to forest plantations in vietnam. *Forests*, 12:1301, 9 2021.
- [63] Nur Arif Rohman and Rahmat Safe'i. Health assessment of tahura banten as an effort to protect biodiversity. In *AIP Conference Proceedings*, volume 2563, page 080004. AIP Publishing LLC, 2022.

- [64] Andri Pranolo and Siti Muslimah Widyastuti. Simple additive weighting method on intelligent agent for urban forest health monitoring. pages 132–135. IEEE, 10 2014.
- [65] Zhenzhao Xu, Qijing Liu, Wenxian Du, Guang Zhou, Lihou Qin, and Zhen Sun. Modelling leaf phenology of some trees with accumulated temperature in a temperate forest in northeast china. *Forest Ecology and Management*, 489:119085, 2021.
- [66] Susanne Wiesner, Gregory Starr, Lindsay R Boring, Julia A Cherry, Paul C Stoy, and Christina L Staudhammer. Forest structure and composition drive differences in metabolic energy and entropy dynamics during temperature extremes in longleaf pine savannas. *Agricultural and Forest Meteorology*, 297:108252, 2021.
- [67] Timothy R Duclos, William V DeLuca, and David I King. Direct and indirect effects of climate on bird abundance along elevation gradients in the northern appalachian mountains. *Diversity and Distributions*, 25(11):1670–1683, 2019.
- [68] Benjamin D Miller, Kelsey R Carter, Sasha C Reed, Tana E Wood, and Molly A Cavaleri. Only sun-lit leaves of the uppermost canopy exceed both air temperature and photosynthetic thermal optima in a wet tropical forest. *Agricultural and Forest Meteorology*, 301:108347, 2021.
- [69] Xiaowen Ge, Jiaojun Zhu, Deliang Lu, Danni Wu, Fengyuan Yu, and Xiaohua Wei. Effects of canopy composition on snow depth and below-the-snow temperature regimes in the temperate secondary forest ecosystem, northeast china. Agricultural and Forest Meteorology, 313:108744, 2022.
- [70] Mingxin Yang, Peng Gao, Ping Zhou, Jiaxing Xie, Daozong Sun, Xiongzhe Han, and Weixing Wang. Simulating canopy temperature using a random forest model to calculate the crop water stress index of chinese brassica. *Agronomy*, 11(11):2244, 2021.
- [71] Marius G Floriancic, Scott T Allen, Raphael Meier, Lucas Truniger, James W Kirchner, and Peter Molnar. Potential for significant precipitation cycling by forest-floor litter and deadwood. *Ecohydrology*, 16(2):e2493, 2023.
- [72] Andy D Beaton, Robert A Metcalfe, James M Buttle, and Steven E Franklin. Investigating snowpack across scale in the northern great lakes–st. lawrence forest region of central ontario, canada. *Hydrological Processes*, 33(26):3310–3329, 2019.
- [73] Ravindra Dwivedi, Joel A Biederman, Patrick D Broxton, Kangsan Lee, and Willem JD van Leeuwen. Snowtography quantifies effects of forest cover on net water input to soil at sites with ephemeral or stable seasonal snowpack in arizona, usa. *Ecohydrology*, 16(2):e2494, 2023.
- [74] Benjamin Bouchard, Daniel F Nadeau, and Florent Domine. Comparison of snowpack structure in gaps and under the canopy in a humid boreal forest. *Hydrological Processes*, 36(9):e14681, 2022.
- [75] Achut Parajuli, Daniel F Nadeau, François Anctil, and Marco Alves. Multilayer observation and estimation of the snowpack cold content in a humid boreal coniferous forest of eastern canada. *The Cryosphere*, 15(12):5371–5386, 2021.
- [76] Michaela Teich, Andrew D Giunta, Pascal Hagenmuller, Peter Bebi, Martin Schneebeli, and Michael J Jenkins. Effects of bark beetle attacks on forest snowpack and avalanche formation–implications for protection forest management. *Forest Ecology and Management*, 438:186–203, 2019.
- [77] L Marryanna, S Noguchi, Y Kosugi, K Niiyama, M Itoh, T Sato, S Takanashi, S Siti-Aisah, and K Abd-Rahman. Spatial distribution of soil moisture and its influence on stand structure in a lowland dipterocarp forest in peninsular malaysia. *Journal of Tropical Forest Science*, 31(2):135–150, 2019.
- [78] Shikha Prasad and Ratul Baishya. Interactive effects of soil moisture and temperature on soil respiration under native and non-native tree species in semi-arid forest of delhi, india. *Tropical Ecology*, 60:252–260, 2019.
- [79] Alejandro Monsiváis-Huertero, Juan Carlos Hernández-Sánchez, José Carlos Jiménez-Escalona, José Mauricio Galeana-Pizaña, Daniel Enrique Constantino-Recillas, Aura Citlalli Torres-Gómez, Ramata Magagi, Kalifa Goïta, and Stéphane Couturier. Impact of temporal variations in vegetation optical depth and vegetation temperature on l-band passive soil moisture retrievals over a tropical forest using in-situ information. *International Journal of Remote Sensing*, 41(6):2098–2139, 2020.
- [80] Paul D Sewell, Sylvie A Quideau, Miles Dyck, and Ellen Macdonald. Long-term effects of harvest on boreal forest soils in relation to a remote sensing-based soil moisture index. *Forest Ecology and Management*, 462, 2020.
- [81] Rafael Coll Delgado, Henderson Silva Wanderley, Marcos Gervasio Pereira, André Quintão de Almeida, Daniel Costa de Carvalho, Douglas da Silva Lindemann, Everaldo Zonta, Sady Júnior Martins da Costa de Menezes, Gilsonley Lopes dos Santos, Romário Oliveira de Santana, et al. Assessment of a new fire risk index for the atlantic forest, brazil. *Forests*, 13(11):1844, 2022.

- [82] Felipe Taliar Giuntini, Delano Medeiros Beder, and Jo Ueyama. Exploiting self-organization and fault tolerance in wireless sensor networks: A case study on wildfire detection application. *International Journal of Distributed Sensor Networks*, 13(4):1550147717704120, 2017.
- [83] Adisorn Lertsinsrubtavee, Thongchai Kanabkaew, and Sunee Raksakietisak. Detection of forest fires and pollutant plume dispersion using iot air quality sensors. *Environmental Pollution*, 338:122701, 2023.
- [84] Faroudja Abid and Nouma Izeboudjen. Predicting forest fire in algeria using data mining techniques: Case study of the decision tree algorithm. In Advanced Intelligent Systems for Sustainable Development (AI2SD'2019) Volume 4-Advanced Intelligent Systems for Applied Computing Sciences, pages 363–370. Springer, 2020.
- [85] Ali Karouni, Bassam Daya, and Pierre Chauvet. Applying decision tree algorithm and neural networks to predict forest fires in lebanon. *Journal of theoretical and applied information technology*, 63:282–291, 2014.
- [86] Sonia I Seneviratne, Thierry Corti, Edouard L Davin, Martin Hirschi, Eric B Jaeger, Irene Lehner, Boris Orlowsky, and Adriaan J Teuling. Investigating soil moisture–climate interactions in a changing climate: A review. *Earth-Science Reviews*, 99(3-4):125–161, 2010.
- [87] Dara Entekhabi, Eni G Njoku, Peggy E O'Neill, Kent H Kellogg, Wade T Crow, Wendy N Edelstein, Jared K Entin, Shawn D Goodman, Thomas J Jackson, Joel Johnson, et al. The soil moisture active passive (smap) mission. *Proceedings of the IEEE*, 98(5):704–716, 2010.
- [88] Luca Brocca, Stefan Hasenauer, T Lacava, F Melone, T Moramarco, W Wagner, W Dorigo, P Matgen, J Martínez-Fernández, P Llorens, et al. Soil moisture estimation through ascat and amsr-e sensors: An intercomparison and validation study across europe. *Remote Sensing of Environment*, 115(12):3390–3408, 2011.
- [89] Vinay Dubey, Prashant Kumar, and Naveen Chauhan. Forest fire detection system using iot and artificial neural network. In *International Conference on Innovative Computing and Communications: Proceedings of ICICC* 2018, Volume 1, pages 323–337. Springer, 2019.
- [90] Scott Kelleher, Casey Quinn, Daniel Miller-Lionberg, and John Volckens. A low-cost particulate matter (pm 2.5) monitor for wildland fire smoke. *Atmospheric Measurement Techniques*, 11(2):1087–1097, 2018.
- [91] Valda Araminienė, Pierre Sicard, Alessandro Anav, Evgenios Agathokleous, Vidas Stakėnas, Alessandra De Marco, Iveta Varnagirytė-Kabašinskienė, Elena Paoletti, and Rasa Girgždienė. Trends and inter-relationships of ground-level ozone metrics and forest health in lithuania. *Science of the Total Environment*, 658:1265–1277, 2019.
- [92] Nicholas M Short. Part of the astrodynamics commons, navigation, guidance, control and dynamics commons, space vehicles commons, systems and communications commons, and the systems engineering and multidisciplinary design optimization commons recommended citation recommended citation short, 2003.
- [93] Jay Fussell, Donald Rundquist, and JA Harrington. On defining remote sensing. *Photogrammetric Engineering* and Remote Sensing, 52(9):1507–1511, 1986.
- [94] Jiaojiao Tian, Thomas Schneider, Christoph Straub, Florian Kugler, and Peter Reinartz. Exploring digital surface models from nine different sensors for forest monitoring and change detection. *Remote Sensing*, 9(3):287, 2017.
- [95] Daniela Stojanova, Panče Panov, Andrej Kobler, Sašo Džeroski, and Katerina Taškova. Learning to predict forest fires with different data mining techniques. In *Conference on data mining and data warehouses (SiKDD 2006), Ljubljana, Slovenia*, pages 255–258, 2006.
- [96] Nathalie Guimarães, Luís Pádua, Pedro Marques, Nuno Silva, Emanuel Peres, and Joaquim J Sousa. Forestry remote sensing from unmanned aerial vehicles: A review focusing on the data, processing and potentialities. *Remote Sensing*, 12(6):1046, 2020.
- [97] Tian-Zhu Xiang, Gui-Song Xia, and Liangpei Zhang. Mini-unmanned aerial vehicle-based remote sensing: Techniques, applications, and prospects. *IEEE geoscience and remote sensing magazine*, 7(3):29–63, 2019.
- [98] Janne Järnstedt, Anssi Pekkarinen, Sakari Tuominen, Christian Ginzler, Markus Holopainen, and Risto Viitala. Forest variable estimation using a high-resolution digital surface model. *ISPRS Journal of Photogrammetry and Remote Sensing*, 74:78–84, 2012.
- [99] Erich Seifert, Stefan Seifert, Holger Vogt, David Drew, Jan Van Aardt, Anton Kunneke, and Thomas Seifert. Influence of drone altitude, image overlap, and optical sensor resolution on multi-view reconstruction of forest images. *Remote sensing*, 11(10):1252, 2019.
- [100] Darren Turner, Arko Lucieer, and Christopher Watson. An automated technique for generating georectified mosaics from ultra-high resolution unmanned aerial vehicle (uav) imagery, based on structure from motion (sfm) point clouds. *Remote sensing*, 4(5):1392–1410, 2012.

- [101] Stuart Krause, Tanja GM Sanders, Jan-Peter Mund, and Klaus Greve. Uav-based photogrammetric tree height measurement for intensive forest monitoring. *Remote sensing*, 11(7):758, 2019.
- [102] Goran Tmušić, Salvatore Manfreda, Helge Aasen, Mike R James, Gil Gonçalves, Eyal Ben-Dor, Anna Brook, Maria Polinova, Jose Juan Arranz, János Mészáros, et al. Current practices in uas-based environmental monitoring. *Remote Sensing*, 12(6):1001, 2020.
- [103] Ana Paula Dalla Corte, Franciel Eduardo Rex, Danilo Roberti Alves de Almeida, Carlos Roberto Sanquetta, Carlos A Silva, Marks M Moura, Ben Wilkinson, Angelica Maria Almeyda Zambrano, Ernandes M da Cunha Neto, Hudson FP Veras, et al. Measuring individual tree diameter and height using gatoreye high-density uav-lidar in an integrated crop-livestock-forest system. *Remote Sensing*, 12(5):863, 2020.
- [104] Xinlian Liang, Yunsheng Wang, Jiri Pyörälä, Matti Lehtomäki, Xiaowei Yu, Harri Kaartinen, Antero Kukko, Eija Honkavaara, Aimad El Issaoui, Olli Nevalainen, et al. Forest in situ observations using unmanned aerial vehicle as an alternative of terrestrial measurements. *Forest ecosystems*, 6(1):1–16, 2019.
- [105] Dameng Yin and Le Wang. Individual mangrove tree measurement using uav-based lidar data: Possibilities and challenges. *Remote Sensing of Environment*, 223:34–49, 2019.
- [106] V Otero, R Van De Kerchove, B Satyanarayana, C Martínez-Espinosa, and M Amir Bin Fisol. Rodila bin ibrahim m, sulong i, mohd-lokman h, lucas r, dahdouh-guebas f. managing mangrove forests from the sky: forest inventory using field data and unmanned aerial vehicle (uav) imagery in the matang mangrove forest reserve, peninsular malaysia. *Forest Ecology and Management*, 411:35–45, 2018.
- [107] Riccardo Dainelli, Piero Toscano, Salvatore Filippo Di Gennaro, and Alessandro Matese. Recent advances in unmanned aerial vehicles forest remote sensing—a systematic review. part ii: Research applications. *Forests*, 12(4):397, 2021.
- [108] David L Hall and James Llinas. An introduction to multisensor data fusion. *Proceedings of the IEEE*, 85(1):6–23, 1997.
- [109] Yun Yang, Martha C Anderson, Feng Gao, Jeffrey D Wood, Lianhong Gu, and Christopher Hain. Studying drought-induced forest mortality using high spatiotemporal resolution evapotranspiration data from thermal satellite imaging. *Remote sensing of environment*, 265:112640, 2021.
- [110] Sean Hartling, Vasit Sagan, and Maitiniyazi Maimaitijiang. Urban tree species classification using uav-based multi-sensor data fusion and machine learning. *GIScience & Remote Sensing*, 58(8):1250–1275, 2021.
- [111] Trevor K Host, Matthew B Russell, Marcella A Windmuller-Campione, Robert A Slesak, and Joseph F Knight. Ash presence and abundance derived from composite landsat and sentinel-2 time series and lidar surface models in minnesota, usa. *Remote Sensing*, 12(8):1341, 2020.
- [112] Benjamin T Fraser and Russell G Congalton. Monitoring fine-scale forest health using unmanned aerial systems (uas) multispectral models. *Remote Sensing*, 13(23):4873, 2021.
- [113] Maitiniyazi Maimaitijiang, Vasit Sagan, Paheding Sidike, Ahmad M Daloye, Hasanjan Erkbol, and Felix B Fritschi. Crop monitoring using satellite/uav data fusion and machine learning. *Remote Sensing*, 12(9):1357, 2020.
- [114] Rajeev Bhattarai, Parinaz Rahimzadeh-Bajgiran, and Aaron Weiskittel. Multi-source mapping of forest susceptibility to spruce budworm defoliation based on stand age and composition across a complex landscape in maine, usa. *Canadian Journal of Remote Sensing*, 48(6):873–893, 2022.
- [115] Minh D Nguyen, Oscar M Baez-Villanueva, Duong D Bui, Phong T Nguyen, and Lars Ribbe. Harmonization of landsat and sentinel 2 for crop monitoring in drought prone areas: Case studies of ninh thuan (vietnam) and bekaa (lebanon). *Remote Sensing*, 12(2):281, 2020.
- [116] Mihai T Lazarescu. Design of a wsn platform for long-term environmental monitoring for iot applications. *IEEE Journal on emerging and selected topics in circuits and systems*, 3(1):45–54, 2013.
- [117] Satoru Miura, Michael Amacher, Thomas Hofer, Jesús San-Miguel-Ayanz, Richard Thackway, et al. Protective functions and ecosystem services of global forests in the past quarter-century. *Forest Ecology and Management*, 352:35–46, 2015.
- [118] Narayan Kayet, Khanindra Pathak, Abhisek Chakrabarty, Subodh Kumar, Vemuri Muthayya Chowdary, and Chandra Prakash Singh. Risk assessment and prediction of forest health for effective geo-environmental planning and monitoring of mining affected forest area in hilltop region. *Geocarto International*, 37(11):3091–3115, 2022.